

Review



Optimal Sizing and Energy Management of Microgrids with Vehicle-to-Grid Technology: A Critical Review and Future Trends

Oussama Ouramdane, Elhoussin Elbouchikhi *D, Yassine Amirat D and Ehsan Sedgh Gooya

ISEN Yncréa Ouest Brest, LABISEN, 20, Rue Cuirassé Bretagne, 29200 Brest, France; oussama.ouramdane@isen-ouest.yncrea.fr (O.O.); yassine.amirat@isen-ouest.yncrea.fr (Y.A.); ehsan.sedgh-gooya@isen-ouest.yncrea.fr (E.S.G.)

* Correspondence: elhoussin.elbouchikhi@isen-ouest.yncrea.fr; Tel.: +33-02-98-03-84-00

Abstract: The topic of microgrids (MGs) is a fast-growing and very promising field of research in terms of energy production quality, pollution reduction and sustainable development. Moreover, MGs are, above all, designed to considerably improve the autonomy, sustainability, and reliability of future electrical distribution grid. At the same time, aspects of MGs energy management, taking into consideration distribution generation systems, energy storage devices, electric vehicles, and consumption components have been widely investigated. Besides, grid architectures including DC, AC, or hybrid power generation systems, energy dispatching problems modelling, operating modes (islanded or grid connected), MGs sizing, simulations and problems solving optimization approaches, and other aspects, have been raised as topics of great interest for both electrical and computer sciences research communities. Furthermore, the United Nations Framework Convention on Climate Change and government policies and incentives have paved the way to massive electric vehicle (EV) deployment. Hence, several research studies have been conducted to investigate the integration of EVs in national power grid and future MGs. Specifically, EV charging stations' bidirectional power flow control and energy management have been considerably explored. These issues index challenging research topics, which are in most cases still under progress. This paper gives an overview of MGs technology advancement in recent decades, taking into consideration distributed energy generation (DER), energy storage systems (ESS), EVs, and loads. It reviews the main MGs architecture, operating modes, sizing and energy management systems (EMS) and EVs integration.

Keywords: microgrids; distributed energy generation; energy storage; optimal sizing; energy management systems; vehicle-to-grid; optimization; energy dispatching

1. Introduction

Last decades have witnessed a significant increase in worldwide energy consumption as shown by Figure 1a, which leads to huge increase in atmospheric pollution as depicted by Figure 1b. Moreover, the increase of fossil fuel cost and the depletion of fossil resources have pushed researchers and engineers in industry to explore and propose a more sustainable energy resources alternatives. Consequently, renewable energy resources (RERs) have been developed to comply with these issues [1,2]. Thanks to their wide availability and non-polluting nature, renewable energies have become the ideal solution to overcome the economic and ecological problems and to ensure electrical energy supply to isolated areas and villages. The advancement and maturity in RERs technologies, such as wind turbines, photovoltaic panels (PV), marine current turbines, biomass, and many others, the advances in power electronics devices, and the development in control and monitoring technologies have made their exploitation more feasible and very profitable [3,4]. The integration of the RERs was done through conventional grids and was connected to the distribution and



Citation: Ouramdane, O.; Elbouchikhi, E.; Amirat, Y.; Sedgh Gooya, E. Optimal Sizing and Energy Management of Microgrids with Vehicle-to-Grid Technology: A Critical Review and Future Trends. *Energies* 2021, *14*, 4166. https:// doi.org/10.3390/en14144166

Academic Editor: José Matas

Received: 2 June 2021 Accepted: 27 June 2021 Published: 10 July 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). transmission grids. As the RERs were connected to the main grid, they are providing a quantity of energy to support the main grid and they have to comply with grid codes for grid integration and to ensure power quality requirements [5].

Given the rapid development of renewable technologies, the need to relieve congestion on existing lines and cables of the electricity grid and to expand overall grid capacity, and the emergency of supplying power to remote areas, researchers have turned their attention to the idea of the microgrid (MG). Distributed energy resources (DERs) are composed of both renewable resources that are sufficiently available in the area where the MG is designed and conventional generators (CGs) including diesel generators (DG), for instance. The energy storage systems (ESS) are paramount in the MGs because they ensure the balance between demand and production, as RERs are intermittent in nature and depend on meteorological conditions. Indeed, when RERs electricity production exceeds the consumption, the ESSs are charging. Contrarily, when RERs production is not sufficient, ESSs are discharging to cover the lack of production. In some circumstances where both renewable resources and ESSs can not fulfil the load demand, conventional generators are used or shiftable loads are disconnected to ensure the energy supply to critical loads. Loads are components of the MG consuming the power generated by DERs available in the area where the MG is designed. The evolution of the load depends on the activity of the people in this area. Consequently, the variation of the consumption is not compatible with that of DERs energy production [6]. This problem is solved mainly using ESSs, but also by the integration of load shedding. This allows some non-critical loads to be shifted outside consumption peak period. Therefore the MG is an autonomous system, which satisfies its consumption by producing its own energy, but it can also operate in a grid-connected mode for energy exchanges and trading with the main grid [7].

MG is defined in [8] as a complex energy system that requires a specific infrastructure, coordination of energy resources and information flows, as well as additional protection and assurance of energy reliability. It is formed by the integration of renewable energy resources, conventional generators, loads and energy storage devices as depicted by Figure 2. MGs can be operated in parallel with the main grid, as a stand-alone power island or in transition between grid-connected and island operation [9]. Moreover, microgrids can be operated as a cluster of interconnected systems with multiple AC and DC microgrids allowing an autonomous and coordinated control, energy management and energy trading.

An MG requires a smart and real-time adjustable management of its three main components. Consequently, an energy management system (EMS) is of paramount importance [10–12]. An EMS that is accurate and able to solve the MG's problems and providing the ideal solution regarding the energy dispatching between DERs, ESS and managing shiftable loads in critical situations. The EMS manages the volatility and intermittency of renewable energy resources and load demand, based on mathematical methods, in order to optimise the overall operation of the MGs in terms of levelized cost of energy, pollution cost, availability and reliability, lifespan of its components, and other objectives while fulfilling several constraints related to grid losses, active and reactive power capacity, energy storage capacity and depth of discharge, and many others. A good energy management has many important impact on the performance of the system throughout its life cycle [13].

The integration of a large-scale renewable energy sources into electricity grids requires battery energy storage. Storage plays an important role in overcoming the intermittency of RERs and ensuring grid frequency stability. When plugged into household sockets, electric vehicles (EVs) can be operated as batteries. It can be charged during off-peak periods and discharged during on-peak periods acting as a DER to support the main grid or MGs. This fact makes EVs act like a load and DER for microgrids, which allows deceasing ESS installed capacity. Therefore, vehicle-to-grid (V2G) technology is considered as one of the most promising key smart grid technologies [14–16]. The aggregated V2G pool designed by a large number of EVs greatly helps the optimal allocation of supply and demand. The owners of these EVs can get an incentive cost [17]. Nowadays, energy management



techniques with V2G in load frequency control and regional EMS are developed in the ubiquitous power grid concept [18,19].

Evolution of energy consumption

Figure 1. Energy consumption and CO₂ emissions worldwide. (**a**) Evolution of energy consumption [20]. (**b**) Evolution of CO₂ emissions [21].

This paper is a critical review and analysis of several works on MGs topologies and components, energy management systems, and electric vehicles integration and its bi-directional energy management. The focus is made on electric vehicles integration into MGs by considering the V2G strategy for energy management in charging stations. Table 1 gives a summary of major papers dealing with the issues of MGs sizing, energy management strategy and V2G concept. Relevant review papers deal with a single or multiple aspects of MGs, which may be generation sources, energy storage systems, sizing, power and energy management systems, or comparisons between AC, DC and hybrid MGs while considering the V2G technology. Few works have brought together all the aspects related to MGs. This article is synthesized in a specific way that provides researchers with a broad summary gathering as much aspects as possible that are related to MGs and EVs integration. The aim of this article is twofold:

- Present different topics and challenges that can be investigated in the field of MGs and V2G technology;
- Provide information on the latest technologies and key locks towards future research topics in the field of MGs.



Figure 2. Microgrid architecture.

The literature offers a lot of information on the different points to be studied in the field of MGs, even if they are treated separately, but the aspects related to V2G are not very present. Today, there is a huge interest in the contribution of V2G in power systems such as load smoothing, energy cost reduction, resilience of MGs, support for the integration of renewable energy, etc. However, there is a little perspective. This article provides researchers with tools to adapt in order to make V2G a reality, giving areas that need to be developed, such as bi-directional chargers, battery degradation management, distributed energy storage capability and different ways of implementing V2G. Before this, the article states the problems associated with V2G technology, and the issues it can involve in distribution grid.

This paper is organized as follows: Section 2 presents the concept of MG by indexing evolutions in the field of renewable technologies, load control, energy storage systems and communication tools. Section 3 describes MGs classification according to power delivery networks, operating modes, and control strategy. Section 4 discusses the aspects related to EVs integration into MGs. Section 5 presents the sizing aspects of MGs. Section 6 deals with energy management systems in MGs. Finally, Section 7 concludes this paper and gives some perspectives for future works.

Ref. Main Contributions In this paper, the optimisation and control techniques are presented and criticized in detail. Optimization methods are [22] classified into several classes to allow the understanding of the advantage and disadvantage of each of them, and their use cases. The main components of the MG and its modes of operation are also presented. The authors presented a comparative study of different EMS in MGs, classified the optimisation objectives, system [23] constraints, solution methods and simulation software used in grid-tied and autonomous MGs. In this work an agricultural investment in the Algerian Sahara is carried out. A hybridization is initiated with local renewable solar energy resource, mainly PVs. This work is limited to the hydraulic pumped storage system, which is [24] effective due to its geological aspects and the subject matter. The proposed management strategy considers temporal solar energy and combines it with pumping and online fuel consumption optimisation. Various configurations and contracts available for the purchase/sale of energy from/to the grid are analysed and [25] compared. The results show the potential to reduce energy costs, pollution and grid reliance. The study presents various functionalities of V2G, such as active power regulation, reactive power support, load [26] balancing, filtering of current harmonics, etc. At the same time, disadvantages are presented such as, battery degradation, communication costs between EVs and the grid, as well as the need for upgraded grid infrastructure. The authors developed models to compute the grid supply capacity of three types of electric vehicles. Several [27] advantages are presented, such as increased stability and reliability of the power grid, reduced costs of the power system, and storage of renewable energy to overcome the intermittency issue. This work shows that there could be a good combination between EVs and power grids, providing an additional [28] storage system. Excess energy produced by RERs can be stored in EVs for driving and released again when consumption demand increases. This work shows that uncoordinated charging causes voltage problems. Authors address this problem by adding a [28] voltage constraint in the optimization model that makes the power bidirectional. In the context of smart control strategies in the Kansai region of Japan, through scenario analysis, the authors evaluate [29] the influence of the implementation of PV, electric vehicles (EV) and heat pumps (HP) in future smart grids. The system also contributes to emission reduction. A flywheel storage system is used in this work. The authors model the autonomous hybrid system between the DG, [30] wind turbine and flywheel. Results show that this storage system can improve the dynamic performance of the hybrid grid in several situations, such as wind power uncertainty, slow response of the DG and load uncertainty. A DC MG based on supercapacity for energy storage is achieved. A control technique is developed for controlling a [31] bi-directional converter that connects the grid with the supercapacity providing the switch and play function. It is demonstrated that supercapacity can reduce the impact of oscillations due to transients in resources and load. The authors designed a multi-layer control allowing a better integration of the DC MG. The objectives are to avoid [32] unwanted injection, to reduce the MG power fluctuations, and to lower the peak load of the network by using a predictive interface system.

Table 1. Main papers about energy management systems of MG.

2. Microgrids Main Components

In this section, the main components of the MGs are presented, which are DERs, loads, sensors and control devices, and communication tools. Regarding DERs, the focus is made on RERs and the energy storage systems (ESS). The main components of the MG are shown by Figure 3.

2.1. DER, ESS and Loads in Microgrids

Table 2 shows examples of sustainable energy systems, energy storage systems and load type in some literature reviews.



Figure 3. Microgrid components.

Table 2. Sustainable energy systems in MGs.

					D	ERs					Load	ls
	CGs		R	ERs					ESSs		Тур	e
	CG	PV	Wind	Marine BIO	BAT	SC	FW	H2	Hydro	Compressed Air	Shedded	Critical
[33]	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark				\checkmark			\checkmark
[31]	\checkmark	\checkmark				\checkmark						\checkmark
[34]	\checkmark				\checkmark		\checkmark					\checkmark
[30]	\checkmark		\checkmark				\checkmark					\checkmark
[35]	\checkmark	\checkmark		\checkmark	\checkmark							\checkmark
[36]	\checkmark	\checkmark	\checkmark		\checkmark							\checkmark
[37]	\checkmark	\checkmark	\checkmark		\checkmark							\checkmark
[38]	\checkmark	\checkmark	$\checkmark\checkmark$		\checkmark							\checkmark
[39]	\checkmark	\checkmark	\checkmark		\checkmark	\checkmark		\checkmark	\checkmark	\checkmark	\checkmark	
[40]	\checkmark	\checkmark	\checkmark	\checkmark			\checkmark	\checkmark				\checkmark

2.1.1. Evolution of RERs Technologies

The integration of RERs depends first and foremost on their maturity. Researchers and engineers obviously opt for mastered technologies. The most mature RERs are:

- Solar energy, which includes three types: photovoltaic panels, solar heating, and concentrating solar power;
- Wind energy, where two types can be distinguished, onshore and offshore wind turbines;
- Marine energy, such as wave energy converters, tidal stream, tidal range [3].

Several articles have focused on the development of these mature renewable resources in terms of sizing, control, energy management and efficiency improvement [4,41,42]. Many other renewable resources can be cited such as hydro energy [43], geothermal energy and bioenergy for electricity and heat, and Biofuels [3]. Moreover, more immature technologies, which are very promising but still in the early stage of development, are investigated by researchers and engineers such as salinity gradient, deep ocean current, ocean thermal energy conversion, etc. [4].

The projects carried out on MGs mainly contain the most mastered RERs at present days. PV is the most mastered renewable energy production technology. Seven types of PVs can be distinguished today, which are photovoltaic solar panels composed of polycrystalline cells, monocrystalline cells, amorphous cells, multi-junction cells, silicon-free photovoltaic solar panels in thin film CIS (stands for the elements copper (C), indium (I) and selenium (S)), air panels and water panels. PVs are the most adopted RERs in MGs, [7,44–46]. In several MG projects, PVs are used alone as renewable energy source to provide the required energy to satisfy the loads. In second place, wind turbines are extensively explored within MGs. Currently, wind turbines available on the market are divided into two large

categories according to the architecture of their electric generator: those equipped with an asynchronous generator (approximately 75% of the market), and those equipped with a permanent magnet synchronous generator (approximately 25% of the market). In several MG projects, production is only based on wind turbines as the renewable resource [47]. Other studies combine the two technologies in order to broaden the range of production, decrease the energy storage systems capacity for investment saving, and sometimes because of the modest availability of the solar and wind resources in the region where the MG is designed. For instance, the research studies in [13,36–40,48] have combined both wind turbines and solar panels to design an MG in order to satisfy the load while fulfilling several constraints. Other renewable energies are used in other research works. Indeed, in [49], the authors have investigated the integration of tidal turbine for islands power delivery. Moreover, in [35], an MG based on biomass and PVs is designed to power a vegetable greenhouse.

2.1.2. Evolution of ESSs Technologies

Given the importance of storage elements in almost all energy fields, especially in MGs, researchers have devoted several works to develop new technologies and enhance the performance of the existing ones. Several storage devices can be distinguished today, which are classified according to criteria of duration, capacity and applications. There are short-, medium- and long-duration storage devices, such as flywheels, batteries and hydrogen and air liquid energy storage, respectively. Large and small energy capacity storage, such as supercapacitors and batteries, respectively. Starting with batteries, they are currently covering a very large part of applications thanks to the availability of several types of batteries, such as Lithium-ion battery (LIB), Sodium-ion batteries (SIB), Redox flow battery (RFB) and so on [50]. They are characterised by their average storage time, and capacity which is not very large, but they can be combined to increase their autonomy. Several studies have been carried out in this field in order to improve lifespan, specific energy, power delivery capacity (specific power), environmental impact and recycling. Due to their high energy density and energy-to-weight ratio, Li-ion batteries are widely used in defence, automotive and aerospace applications [51]. The Ragone plot compares different storage systems according to power density and energy density as shown in Figure 4.



Figure 4. Ragone plot showing specific energy versus specific power for various energy-storing devices [52].

Unlike batteries, electrochemical (EC) capacitors can operate at high charge and discharge rate over almost unlimited number of cycles and allow energy recovery in heavier systems [53]. Flywheel systems are under development and include those with steel flywheel rotors and resin/glass or resin/carbon fibre composite rotors [54]. Hydrogen energy storage is seen as a transition vector by researchers. The surplus energy generated by intermittent renewable energy sources is used for the electrolysis of water in order to produce hydrogen. The hydrogen is then stored as a liquid or compressed gas for later use. The hydrogen can be produced, stored and used to produce energy and water using fuel cell technology during peak-energy demand [55].

2.1.3. Evolution of Loads

Loads represent the second part of the MG after the DERs. Several aspects are dealt with by researchers in order to optimise their consumption, improve their protection systems, but above all to limit the self-discharging over time. In some harsh circumstances, energy consumption can rise abruptly at any time of the day, and the DERs, including the ESSs, may not be able to meet the energy demand. To solve this issue, load shedding can be operated in order to ensure the power delivery to critical loads. Indeed, the idea is to shift some non-critical loads (shiftable) outside the high energy consumption period in order to shave peak demand [56]. On the one hand, loads such as electric water heaters, refrigerators, electric vehicles, washing machines and public lighting can be shifted to maintain active power balance and MG stability. On the other hand, other loads are considered as critical. Indeed, they need to be powered regardless of energy production, such as hospitals, military bases, heating systems and university residences. In [32], the authors have performed a load shedding to ensure energy balance in an MG. Two operating modes have been investigated: in the first mode, the load is critical and must be satisfied at all times, whereas in the second mode, the load is controlled; if it exceeds a certain threshold it is divided into two parts according to some criteria. One part requiring to be powered, and the second one can be shifted later. Load shedding has shown its influence on the whole system, peak loads have decreased, DERs operate less intensively compared to the first study. In [57], authors proposed a model for optimising controllable load shedding for emergency MG control. The model takes into account the efficiency of load losses and the cost of load shedding. The aim is to minimise the load losses and voltage fluctuations of all buses.

2.2. Evolution of Communication Tools

In MGs, communication devices are of paramount importance. Indeed, all control and monitoring units and processors need an efficient data communication system. The aim is to ensure continuous, fast, reliable and accurate data transfer between sensors and controllers without any disruption or disconnection. Communication tools can be expensive, so it is essential to reduce installation costs while maintaining reliable operation [58].

The communication technologies used for data transmission between smart meters and utilities fall into two categories, wired and wireless [59]. In some cases, wired solutions are preferred because they do not have interference problems and their operation does not dependent on batteries, as wireless solutions often do. On the other hand, wireless communications have certain advantages over wired technologies, such as easy connection to difficult or inaccessible sites and low infrastructure costs [59].

The improvement of communication systems has enormously contributed to the development of MGs. In [59], authors show the evolution of smart sensors installed on DERs and smart meters installed in consumers homes. Some very practical communication technologies with and without wires are presented in [59], with their advantages and disadvantages, such as ZigBee, Wireless Mesh, Cellular Network Communication, Powerline Communication and Digital Subscriber Lines. Figure 5 presents a comparison of communication technologies used in MGs operation in terms of data rate and coverage area.





2.3. Evolution of Electric Vehicles

The energy transition emphasizes a double challenge for energy suppliers. On the one hand, they must succeed in widely integrating intermittent renewable energy sources such as solar and wind power without being able to store their production on a large scale. On the other hand, it is necessary to ensure the stability of the grid and the ability to respond instantly to consumers demand including EVs fleet, which is expected to significantly increase in the upcoming years.

Batteries in EVs can help in the management of the energy of MGs since they can be considered as potential energy storage devices [60]. They can, therefore, participate in the energy management of MGs by storing energy in the case where renewable energy production exceeds energy demand (Grid-To-Vehicle, *G2V*) and supplying energy to the grid (Vehicle-To-Grid, *V2G*) during peak demand periods [61], as shown in Figure 6. In order to achieve this revolutionary concept, several parameters have to be studied such as batteries sizing and charging stations sitting and associated power electronics. The idea, therefore, is to consider the battery of an EV as an extension of the distribution grid, a distributed energy storage devices from which the electricity supplier can draw energy at any time.



Figure 6. Vehicle to Grid concept.

V2G concept presents many advantages. For example, a householder can use the energy stored in the EV's battery to power different appliances during peak hours or time slots when electricity prices are high; and the EV battery can be charged later at night (during periods of low electricity prices or low electricity use), when the electricity provider offers the lowest prices. This is possible only in some countries where the price of electricity varies during the day (dynamic pricing) and may be considered in the implementation of MGs in the near future. In the same way, the flexibility of the V2G will allow batteries to be charged during renewable energy production time slots making the charging stations more environmentally friendly. Moreover, V2G makes the electricity from renewable resources available when the supply offered by solar or wind power is interrupted (during night for solar energy, for instance) [62].

At the network level, the distributed storage capacity made available through V2G is used by operators to respond more effectively to variations in demand. It helps, for example, to absorb peak demand without load shedding, or to compensate for micro-disturbances that can occur when production switches from one energy source to another [62]. In this model, operators pay their customers for the use of their batteries: V2G thus enables the end consumer to reduce their energy expenses. Finally, the vehicle-to-grid contributes to the creation of a smart distribution network: a smartgrid, in which the flows are constantly optimised thanks to measurements taken at each link in the chain. Combined with the technical capabilities of EVs, the smartgrid logic enables islands such as Porto Santo or Belle-Ile-en-Mer to make very concrete progress on the road to energy transition [62]. Some possible applications of the V2G concept ara as follows:

- Voltage and frequency control:
 - Voltage stability is maintained when there is a balance between reactive power supply and demand. EVs act, instantly, to regulate signals that could be separately handled by each EV. A battery charger is used to integrate voltage regulation. The EV charger can manage the charging current in such a way to have the necessary phase angle to compensate for capacitive or inductive reactive power [63]. The charging stations can be managed such that at a low voltage, EV charging stops, and it resumes when the voltage level is high.
 - Frequency stability is maintained when there is a balance between active power supply and demand. In [63], frequency regulation techniques are presented using large cyclic generators, but they are expensive. The rapid charging and

discharging of EV batteries can offer an alternative to frequency regulation [27]. Three types of control to maintain stable frequency exist, which are primary, secondary and tertiary control as described in [64].

- Load balancing and peak power (load management): The bi-directional operation of the *V2G* system can manage the electrical charge, such that in peak hours it discharges, and it charges when demand is low, i.e., during the night and off-peak hours. In [65], a smart charging algorithm is developed to allow peak load management and to shift the consumption curve. Shifting the load can be achieved through load coordination, reducing the impact of the EV fleet on the grid. The objective of the controlled battery charger is to shift the energy demand and level the peak load. In [66], authors suggest that shifting the load curve by peak power control is the cheapest solution for load management.
- Support to renewable energy resources: *V2G* can be a backup source for renewable resources during their low output, providing alternative energy production [25]. Centralized power plants need to reduce their production by decreasing the number of distributed generation units to restore the balance. Conversely, EVs can store excess energy produced by RERs when they are in peak production, and then be discharged and fed into the grid when demand is higher [28]. In [67], the authors show that distribution networks with smartgrids and RERs are much cleaner than other systems and save industries USD 3.58 per vehicle per day.

The V2G system has many advantages, but with the increase of the number of EVs, it could have a direct impact on the dynamics of the power distribution network and system performance by overloading transformers, cables and power supplies. This reduces efficiency and requires additional generator starts, and creates voltage variations and harmonics [68,69]. The resilience of the network, its reliability, the balance between production and consumption, the transmission of electricity from supplier to consumers, all this can be maintained by the ancillary services that are necessary in an electricity system. These ancillary services can be provided by the good quality two-way V2G system, as well as better voltage regulation and frequency control and efficient load management and spinning reserves. Each electric vehicle can be used separately or as part of a cluster [26].

In addition to V2G concept, energy players can develop a storage technology using second-life batteries. When a battery no longer meets the requirements of automotive use, it can, before being recycled, be used in a "second life" for the large-scale storage of electricity. A less restrictive mission and less demanding in terms of energy and power density. Indeed, the battery is considered as much more than a mobility tool. Once its automotive life is over, the battery's residual value is enormous and can be used for other devices that are less stressful than automotive use, such as stationary storage [62].

3. Microgrids Classification

The previous section showed the main components of the MG—the DERs, including ESSs, EVs, loads, and communication tools. These elements are brought together to give a reliable MG that is able to ensure the production/consumption balance. The design of the MGs follows an architecture that depends on the aplication, and MGs can be classified according to three criteria into three main classes. The first criterion is the type of power which can be AC, DC or Hybrid. The second criterion is the operating mode, which can be isolated or connected. Finally, the last criterion is the type of control, which can be centralized or decentralized. Figure 7 shows the classification of MGs based upon these criteria.



Figure 7. Microgrids classification.

3.1. Type of Power

As soon as electricity has to be transported between production sites and consumption sites, alternating current became the leading technology. Its strength lies in the possibility of using transformers to bring it to high voltages in order to make its transport easier and more profitable. Generators, both AC and DC, produce electricity at relatively low voltage levels, which is impractical for long-distance transmission, or else it results in huge and costly energy losses. For this purpose, the use of the step-up transformer has become necessary, involving the use of AC [70]. Alternating current also offers a high degree of interconnection flexibility to form a strong grid network, which makes electricity supply more reliable. In the early days of electricity, the issue of reliability of supply was a major concern. Then, as production came closer to consumption, the problem of massive transmission over long distances became less acute [71]. The main advantage of the alternative: the flexibility of connecting loads and production on the power path. This advantage is particularly important if the transport corridor passes through a densely populated area and if production facilities are located along the route. On the other hand, the alternative is costly: the system described above is very expensive since, in reality, it requires the construction of an all-electric infrastructure from start to end [70,72].

The major disadvantage of AC is the presence of reactive power. When a line is loaded below its nominal power, reactive power is produced; without the addition of compensation equipments, the voltage can rise disproportionately. Conversely, a line loaded above its nominal power consumes reactive power and causes the voltage to drop sharply. From a reliability point of view, it is necessary to build an AC transmission line in the form of interconnected sections, combining series and shunt compensation, in order to carry the maximum power at all times [71]. The fields of control and protection also have their own particular constraints. It must be possible to eliminate earth faults on one phase without opening the circuit breakers of all three phases. The difficulty lies in the high capacitive current, generated by the healthy phases, flowing through the fault. This can be overcome with tuned reactances that minimise the induced current [71].

Direct current present several advantages. Indeed, DC is easier to transport over long distances. The converter stations (rectifier-side and inverter-side) can very quickly regulate the voltage/current values and thus the power flow. The phase shift between the transmitting and receiving ends is insignificant if the only connection is direct current. In fact, the connected networks can even operate asynchronously since direct current has no phase angle and is independent of frequency. In addition of that, the emergence of new DC systems such as PV and other household appliances and the reduction in battery prices have paved the way for DC microgrids [70]. Another advantage of direct current lies in its low cost of transporting very high power over long distances, with minimal losses (5% for 2000 km). Not to mention that it requires fewer lines and a smaller footprint on the environment: the transfer of 12,000 MW can be satisfied by two DC lines at 800 kV, whereas the alternative would require eight lines [71].

Nevertheless, the combination of a massive power transmission over an economical high-voltage direct current link and, in parallel, a lower-voltage alternating current network could, in many cases, offer an optimal solution combining low cost and high flexibility, as well as the possibility of supplying the consumer throughout the entire journey. However, this does not come without some technical problems: disturbances in the DC transmission will often cause the AC link to break as the phase angles increase [71].

3.2. Type of Control

As mentioned before, the key component of the deployment of a MG is the use of an optimal and efficient energy management, which can be provided by a centralized or decentralized control system. In the centralized control system, the control is provided by a single control unit, which manages the entire MG. It receives internal information from the system (from sensors), and external information (climate data) and keeps tracking the energy market, especially the price evolutions, and generates a solution taking into account all these data at the same time. Therefore, the computed solution is feasible and optimal, and meets the system requirements, because it is generated according to all the system constraints [73]. However, collecting all the system data in a single control unit results in too many errors, too many variables and equations. The system constraints are accumulated and the large number of variables makes optimization very difficult, very slow, and the system becomes more complicated to solve. This type of control requires a strong computational approach and especially a fast solver to generate the solution in real time [74]. Another disadvantage of this type of control is that the optimization may fail due to a problem in one component of the system, resulting in a loss of management and the shutdown of the whole system.

The second possible control structure is decentralized control. It is developed to overcome the shortcomings of centralized control. The advantage of this control is that each unit takes care of its subsystem locally, therefore it will have fewer tasks to perform [75]. The local unit collects the measurements related to its subsystem and analyses the query it receives from the global unit. Initially, this unit generates a local solution taking into account only the constraints of its subsystem, which reduces the number of variables and thus the computation time. After calculating the local solutions, the central unit uses these results to find the optimal solution for the complete system, without taking into account all the constraints of the system, nor the measurements made at the level of the sensors of the subsystems. The computation time is short, and the approaches used are more simplified, and sometimes they are linear, and do not require strong software for computation [73].

Decentralized control, despite its advantages, has some drawbacks. It requires several local control units, more sensors, more communication tools, and other facilities, and thus a more expensive investment. Moreover, the solutions computed by the local units may be rejected by the central unit, as they are not feasible and do not satisfy all the constraints of the system [73]. In [76], the two aspects are combined for voltage control from power electronic converters. Several advantages are presented: adaptability to failures and the possibility of overcoming them (fault-tolerant approach), uncomplicated implementation, good power and energy management and thus good efficiency.

3.3. Type of Operation

A MG can operate in a connected mode (ON-grid) or in an islanded mode (OFF-grid) to increase the reliability of the power supply by disconnecting from the grid in case of

a grid failure [77]. Rural electrification and small islands are some examples of islanded MGs. Their capacity is in the range of a few watts to 10 MW. Larger islands, industrial sites, military sites, hospitals, data centers, and existing distribution networks are some examples of island and connected MGs, their power ranges from 500 kW to 10 MW [78]. Large transmission and interconnection networks carry very large quantities of energy over long distances, they are voltage regulated to reduce current (losses), voltage levels are between 225 kV and 400 kV and can go up to 1000 kV. Then, there are the regional distribution networks, the voltage level varies between 63 kV, 90 kV and 225 kV. Finally, the distribution networks serve the final consumers at medium or low voltages. MGs, as their definition indicates, contain loads and they are designed in populated areas, so they have lower voltage levels than the large transmission and distribution networks, they are connected to the low voltage (sometimes medium voltage) distribution networks, where the voltage level varies from 400 V to 20 kV.

Connected MGs have a point of interconnection (called a point of common coupling) with the main grid, allowing them to operate in both modes. MGs with this characteristic of operating in both modes must switch from one case to the other in a smooth and transparent manner [79]. This ability to switch modes makes the system more resilient, and valuable in the event of failures [80]. These aspects of switching from one mode to another are not present in MGs operating only in island mode, and therefore, less constrained in the EMS. An MG has an infrastructure that communicates information between the central control unit and its other measurement and control parts, but in general, an infrastructure is installed in parallel between all these parts and the main grid operator [6].

In a grid-connected mode, voltage and frequency are controlled by the main grid control unit, making the management of the MG simpler, as it becomes like a secondary component of the main grid, and therefore only satisfies the instructions of the main unit. The only parameters to be controlled are those of the DERs, the loads, and the ESSs [81]. In contrast to this mode, the island mode is more complex; in addition to the above parameters, the voltage and frequency must be controlled at all components of the islanded MG. Among the major disturbances encountered in MGs, the involuntary switch from connected mode to island mode, which is added to other known disturbances, such as the loss of one or more DERs and short circuits [6].

4. Electric Vehicles Integration into Microgrids

The majority of EVs are parked mostly all the time. To cope with the high penetration of intermittent RERs, the concept of *V2G* is considered as a promising solution. EVs have many advantages, but they also have disadvantages, among them their high energy consumption. Indeed, charging an EV is equivalent to the consumption of a house in Europe or the United States for a day. So, the attractiveness of EVs is only guaranteed if the penetration of RERs is high [82]. Indeed, if EV integration is not followed by high RER penetration, an energy shortage is created. Nevertheless, the power system has to overcome this load increase, and negative consequences are expected, such as ageing of power system DERs, degradation of ESSs, or even lack of energy at charging stations. EV charging stations can be categorized into two main classes as depicted by Figure 8 [83]. EV charging, which is the cheapest but requires tracking and management [84]. Staggered charging times can help save the operation of the electrical system, and do so safely. The practice of load time staggering can save up to 5–35% of critical capital costs and reduce losses by up to 40% [82].





4.1. Network and EVs Constraints

The increase in the number of EVs in the grid has some impacts on generation and transmission, but especially on the distribution system because of the location of charging stations and EVs during the day. Loads and power losses are the most important parameters to control in a *V2G* system. For the loads it is necessary to reduce or eliminate fluctuations, and to improve the power quality in terms of voltage stability, three-phase unbalance, harmonics, etc. The planning of the network, its efficiency, reliability, benefits for the EV owner, and comfort, are addressed by the proposed model in [85,86]. Therefore, the works dedicated to *V2G* concepts is mainly focused on the optimisation of harmonic pollution and load fluctuations. Despite the significant attention from researchers, the *V2G* technology is still a new concept in smart MGs. The most studied aspects related to this concept are:

- Load balancing;
- Harmonic elimination;
- Load diversion to avoid spikes;
- Optimisation of the operational cost of the system;
- Improvement of load factors;
- Minimising emissions;
- Encouraging RERs integration.

To date, the economic feasibility as well as the overall structure have been addressed in several papers, however little work has been conducted on the implementation of the V2G concept. Willett in [27] has performed some work on the calculation of system capacity, and has also dealt with network stability problems in the case of very large integration of

RERs. In [87], Dirk has investigated the impact of storage by EVs and shown that their capacity can be sufficient and may replace static ESSs. However, the location of EVs and charging stations remains a concern to be improved. Another technique is presented in [88], which uses EV storage to overcome peak loads. According to [89], the V2G implementation can be divided into four categories based on the different characteristics and methods of powering EVs:

- Centralized implementation: the network controls all the system including the EVs. This category is applicable for systems with very large charging stations.
- Stand-alone implementation: it does not allow for unified control. However, the features are flexible, convenient and not dependent on time or location.
- Implementation in an MG: The EVs are integrated to perform *V2G* just in one area. There is no exchange with the main network. It can ensure self-sufficiency in this area.
- Battery replacement: reducing the number of storage batteries by integrating more EVs into the system. The interest is to minimize the initial investment and maintenance cost.

The issues to be addressed by the *V2G* concept, can be summarized as [90]:

- The distribution of charging stations, and EVs.
- The management of EV charging, which remains complicated due to the unpredicted use of EV by users.
- The bidirectional charging, requiring a low-loss charger.
- The impact of the V2G concept on batteries performance (battery degradation, reduced life cycle, etc.).

The literature indicates that the initial source of the two major problems associated with V2G, which are harmonic pollution and load fluctuations, is the bidirectional charger [90]. Indeed, conventional chargers operating in one direction only, has problems with harmonics. Therefore, the new chargers between the grid and the EVs must not only work in both directions, but they must also eliminate harmonics. The objective is to find the right topology to design chargers that can add value to the V2G concept, such as improved efficiency, economic optimisation, harmonic elimination, and reduced number of static ESSs, while remaining stable and resilient to external disturbances [90].

Regarding chargers, two levels of charging are mainly focused on: slow charging, which lasts a few hours, and is done at home or at work. Fast charging, which lasts a few minutes, and is done in public places [82]. The current networks are AC, so to supply the batteries of EVs which are DC, a transformer, rectifier and chopper must be added. This chain of converters increases the investment cost as well as the heating losses. The efficiency of the system may decrease especially for fast charging, which requires high current and voltage, the size and volume of these converters are not negligible.

The research activities for control in the V2G system are based on several objectives: optimisation of the operation to reduce cost, reduction of losses, minimisation of load variation, and good energy planning for the EVs. Solutions to these issues are separated into two categories:

- Use smart approaches to manage the operation of all EVs. An algorithm is proposed in [91], which showed some drawbacks as it makes the problems more complicated and does not take into consideration the proprietary side of the EVs.
- Use a decentralised model as in [17]: the principle is to put an intermediate system. Each area containing a number of EVs is managed by a management system, so the main network will not handle the details of each EV.

The research carried out has not taken into consideration all the actors in the *V2G* system. The users of EVs are one of the most important actors in this concept, they must be willing to be part of this energy management program. Currently all the work requires that users accept the developed network management, while in reality this management may not satisfy all users. When energy is exchanged between the EV and the grid, whether in one direction or the other, each EV is seen separately from the other EVs. This is because

17 of 45

this exchange is not done in the same way for all EVs. It changes from one EV to another. It depends on the real time and future status of the EV, which is mainly the battery charge level, its location with respect to the charging stations, and the distances the vehicle will travel, and in some countries it depends on the electricity price [90].

The uncoordinated loading of EVs is at the root of these problems [90]. As the network gets larger, the number of EVs increases, making the system more complex, due to the randomness of the EV as explained. Research that has attempted to address this problem includes pilot studies at the technical level, ranging from studying the influence of the randomness of vehicle charging to measuring the benefits of discharging energy from the vehicle to the grid [90]. Another key point that is a key lock for the development of the *V2G* concept lies in need for real time information, which facilitates the forecasting of consumption and pricing. For this purpose, the advanced metering infrastructure is indexed as a promising solution, because it allows to connect both the smart metering unit and the real-time communication medium [82].

Finally, there is not much systematic work to solve these problems, and even less for energy pricing in both directions. Solutions need to be found that encourage users to participate in this concept, offer more benefits, keep users' EVs in good condition (the vehicle's battery) and especially provide an interest in the energy price between charging and discharging (price incentives).

4.2. Power Quality Standards

The power quality that medium voltage and low voltage networks must provide to users is referred to in several standards [92]. Both the International Electrotechnical Commission (IEC) and the Institute of Electrical and Electronics Engineers (IEEE) define power quality standards such as the IEEE-519 and IEC 61000-4-30. Moreover, the European standard EN 50160 describes the phenomena that can degrade the power quality of the electrical network (UTE 2000). The power quality disturbances include the variation of frequency, amplitude of the supplied voltage, voltage dips, short and long interruptions, temporal and transient overvoltages between phases and earth, harmonic and inter-harmonic voltages, transmission of information signals on the network and unbalance of the supplied voltage. The standards for charging infrastructure are set out in Figure 9. These standards provide the requirements for charging infrastructure in terms of electromagnetic compatibility (EMC), energy storage, low-voltage safety, and communication between EVs and the power grid. Among the standards applicable to EVs, ISO TC22 Road vehicles regulates everything that concerns the EV as a means of transport (braking, power, energy consumption, frontal and lateral impacts, safety, etc.). The applicable standards on fuses and circuit breakers are EU2006/95EC, EU2004/108/EC, IEC TC64 and IEC TC 57. The storage cell standards that apply to batteries are EN 1175, EN 50272 and J2464, in addition to those related to their safe handling such as ISO 12405, ISO 6469, IEC 61982 and J1495.

Three standards are directly related to EV charging stations [93], which are:

- IEC 61851, which regulates:
 - The non-isolated charging system for electric vehicles that defines the charging modes;
 - The types of outputs from the charging stations (both AC and DC) to the EV;
 - Safety requirements for EMC and for the connector socket.
- The connectors for the EV are designed according to IEC 62196 and SAE J1772 [83].
- Communications between the charging station and the EV are regulated by SAE *J*2847. This regulates the communication between the EV and the power grid to achieve energy transfer. The ISO 15118 standard defines the characteristics of the communication interface between the vehicle and the electrical network [94].



Figure 9. Diagram of the standards surrounding the connection of the EV to the electrical grid [83].

4.3. Main Contributions Regarding V2G

The V2G concept requires three elements to be realised: The vehicle must be connected to the network for energy transfer, communication between the vehicle and the network to receive the command and the measuring devices for the command [95]. The V2G system can be used with hybrid EVs, fuel cell EVs or pure battery EVs [95]. In [96], an analysis of hybrid vehicles is made for different consumption cases: peak consumption, base consumption, spinning reserve and regulation services.

Several control works have been proposed for a safe and optimal operation of the *V2G*. Various authors around the world have studied the challenges posed by the *V2G* and the different management techniques. In [97], the impact of the bi-directional charging of the Li-ion cells of EVs was proposed to determine their performance. Paper [98,99] discussed battery technology and *V2G* technology policy. A methodology is developed to manage battery degradation and extend the life of the battery used in EVs. In 2015 Habib et al. conducted a study on the impact of EVs on the electricity grid by benchmarking the charging strategy of an EV in addition to *V2G* technology. They also stated that the charging strategy and aggressiveness of the vehicle could make *V2G* technology economically viable [26]. In [100], Birnie conducted a study examining parking in the United States and New Jersey and determined the driving needs that could be met by solar power only in summer using a simple approach. A similar study performed in [101]. Indeed, authors considered the possibility of providing parking in workplaces in the USA, Ohio, Columbus, Los Angeles, and CA. The system showed that it is efficient in terms of billing and carbon dioxide emissions.

Many studies have considered EV fleets at another urban or regional level. Among these studies is the project carried out in the Kansai region of Japan [29], which uses a smart charging method, combining one million EVs with one million heat pumps to reduce excess solar energy by about 3 TWh. The research work in [102] showed that batteries have almost no impact on the grid because they the capacity is small.

There are many socio-technical obstacles to the development of V2G due to its wide deployment [103]. In [27], Kempton and Tomic conducted a research to assess the economic aspects of V2G. They expressed the battery energy life as a function of several parameters, such as battery cycle life, battery capacity and density. In [104], Ekman examined the benefits of EV fleets for the high penetration of wind power in Denmark.

5. Microgrid Sizing

The type of power is the first criterion to be chosen before designing the MG. Indeed, an AC MG does not have the same operating conditions as a DC MG. Therefore, the two microgrids type do not have the same architecture or energy management strategies. Moreover, MG sizing involves other aspects that should be handled, which include conversion efficiency, transmission efficiency, control, calculation methods, the need for adequate loads and finally the protection systems [105]. A general structure of AC and DC microgrids with renewable resources are provided in Figure 10.

MGs contain AC and DC components at different voltage levels, which requires the addition of converters between each component and the rest of the system. In AC MGs, PVs, batteries, DC loads and all other DC components are connected to DC/AC converters, while other AC components, such as wind turbines, AC loads and the main grid (AC in general) are connected through AC/AC converters so that they are at the same voltage level and frequency as the overall system. In contrast, in DC MGs all components are connected to a fixed voltage DC bus. The DC components pass through choppers (DC/DC converters) to have the same voltage as the DC bus. The other AC components are connected to rectifiers to rectify their voltage to DC voltage.

The role of the converters is not only in the conversion of voltage, they are the point of action for the control of the whole system. The guidelines generated by the EMS are sent to the converters to achieve the optimal operating point. Power electronic devices keep the voltage constant to ensure the safety of the network, switch the DERs ON or OFF, and vary the current intensity at the output of the RERs according to the power reference required to achieve the generation/consumption balance.

5.1. Components Sizing, MG Siting and Operation Scheduling

Starting with the first point, which is the choice of DERs and the location of the MG. The designers of the MG must obviously choose a location that is suitable to be the one dedicated to energy production. It must be rich in renewable resources, and easy to access, and above all the power lines of the network must serve the customer areas [106]. Once the site is appropriately chosen, MG designers must select the best available DERs to meet the loads in a given area [106].

The energy production and storage equipment must be designed in such way that it is capable of supplying the load at any time and especially at any level. For this reason, the DERs must be sized according to peak demand and profitability criteria. Adding to the DERs sizing, the choice of fuel types adapted to the DGs, which is a critical issue with regard to the profitability and reliability of the system. A good initial investment may cost a little, but it will play a very important role afterwards, not only in satisfying the load, but also in earning more than one has invested. For this first point, several objectives need to be achieved: high cost-effectiveness, high profitability, low environmental impact and high reliability, but also low power losses [106].

Scheduling focuses on the planning of available resources, such as DGs, RERs and ESSs. The objective of scheduling is to minimise operational costs, environmental impact and maintenance cost while satisfying the load. Using different optimisation techniques for one or more objective functions, optimal operational conditions are dealt with for different micro-grid configurations in [106].

Optimal sizing is a very important factor for a reliable low power supply. Several algorithms exist for the computation of the size of power systems. Optimal sizing techniques are divided into three classes [107]: classical approaches, modern approaches and computational software. Currently, modern approaches are more suitable than classical approaches, due to their ability to solve and optimise complex models. In addition, these approaches are combined to formulate a hybrid algorithm to provide more promising proposals [108,109]. Calculation software such as HOGA, IHOGA and HOMER are the most widely used in the literature for MGs design. The simplest techniques are the deterministic approaches, citing: analytical, numerical, iterative, graphical construction methods.

Heuristic and metaheuristic algorithms are also used for sizing, for example: the firefly algorithm, particle swarm sizing, the grey wolf technique, the Cuckoo search approach and genetic algorithms [110–115]. The only drawback regarding these approaches is it does not allow intuitive selection of network components. They are used under some assumptions as discussed in [116,117]. The mentioned tools are used to perform comparative studies to investigate the sensitivity of the design results in [118].



Figure 10. Renewable resources-based AC and DC microgrids architecture. (**a**) AC microgrid topology. (**b**) DC microgrid topology.

5.2. AC Microgrids Sizing

For various advantages, the AC MG has gained attention over the DC MG in the emergence of smart grids. AC MGs allow the use of RERs like DC MGs, but also non-renewable energies, such as diesel generator. The initial investment of an AC MG is more cost-effective in terms of the price of the devices, as they are available in large quantities and of good quality for all sizes of MGs, all over the world, from all companies. To connect to another conventional grid, AC MG is more recommended, as most grids are conventional and sometimes contain only fossil resources [119]. AC MG is reliable and has proven to be resilient in both populated cities and rural areas.

The presence of outages and disturbances does not have a great impact on a well-sized AC MG, as it can easily scale up production to overcome this problem. In islanded mode, the AC MG can operate with fewer grid-related constraints. Although AC systems are mature and controllable when installing overhead power lines, the management system remains complicated in an MG context, especially in terms of frequency control, and requires the designers to integrate sensors on all devices so that they are all synchronized [119]. In the design of MGs, the safety aspect is of great importance. In AC MGs sizing, the most effective protective devices are: (i) overcurrent relays, (ii) reclosers, (iii) disconnectors, (iv) miniature circuit breakers and (v) fuses. Currently, in low voltage distribution systems, the most commonly used protection schemes are: (i) fuse saving schemes, (ii) fuse blowing schemes, (iii) instantaneous reclosing and (iv) delayed reclosing [120].

5.3. DC Microgrids Sizing

The remarkable return of DC devices has favoured the design of DC MGs. However, instead of installing several converters, which will affect the accuracy of sizing, it will be better to design a DC MGs to reduce the number of converters, and therefore decrease losses. When sizing a DC MG, there will be fewer converters and therefore fewer loss calculations to take into account. On the other hand, DC/AC converters are added, but due to the high infiltration of DC equipment, the elimination of AC/DC converters is more important than the addition of DC/AC converters. For the same amount of power, a DC system requires smaller line cross-sections, as they carry a lower current than AC systems, due to the power factor. By minimising the current, the losses will also be reduced [121].

In a DC system, even if it does not need reactive energy, some of its components do. The AC equipment must therefore be properly sized to compensate for the reactive energy. Protective elements remain a problem in DC MGs, as they are immature and too costly. In [122], Bosch presented a new design of DC MGs with PVs as DERs to power DC loads. The aim is to design a reliable system with minimum converters. The results show that this new concept significantly increases the energy efficiency, which would be lower when using an AC MG for the same PVs and load.

6. Energy Management Systems

An Energy Management System (EMS) is a group of information technology tools used by power grid management units to control, monitor and optimise the performance and cost of power generation and the power grid while ensuring its continued operation [123]. Figure 11 depicts the main EMS architecture providing an insight on EMS inputs (weather forecasts and energy market), components (DERs, ESS, EVs, Loads, etc.), sensors, control, protection and monitoring devices.

The International Electrotechnical Commission, in IEC 61970, defines an EMS as "a computer system comprising a software platform providing basic support services and a set of applications providing the functionality necessary for the efficient operation of power generation and transmission facilities so as to ensure adequate security of energy supply at minimum cost" [22]. The problem of MG optimisation generally encompasses the following points:

- Minimizing the operating costs of the micro-grid;
- Maximising the output power of the generators at a given time;



Minimising environmental costs.



Figure 12 describes all the aspects related to energy management systems in MGs. In general, an EMS problem leads to a constrained non-linear mixed-integer optimization problem, which can be relaxed to linear optimization problem. It can be solved using linear or non-linear methods, which can be implemented using available software such as Matlab, HOMER, etc. Several cost functions can be considered such as investment cost, operating and maintenance costs, losses cost, emissions cost, etc. The objective is to find the optimal values for decision variables that allow minimizing the cost function considering actual measured data (weather forecast, energy market, load demand, etc.) and fulfilling the constraints of the system. Specifically, the following subsections present a discussion on the objective functions of EMS, electrical network constraints, optimization methods, and software for optimization problem solving and MG optimal planning.



Figure 12. Energy management strategies of MG.

6.1. Microgrid Optimal Operation and Resilience

As discussed in [124], the objective of the EMS is to generate appropriate setpoints for all energy sources, storage and loads, so as to maintain an economically optimised energy mix to meet a certain load demand at a certain point in time. Since renewable energies always change according to climatic data (wind speed and direction, solar irradiation and ambient temperature, etc.), production forecasts and some fast online algorithms are used to define the availability of energy and, finally, to define the signals for optimised power distribution to loads.

As a result, energy management has to find an energy plan for one day or several days in advance. An intelligent energy management system is therefore required to enable short-term planning of energy allocation at minimum costs based on power generation and load demand. The EMS optimises the operation of the MG according to market prices, DGs offers, predicted generation, and expected loads. Based on these data, EMS sends signals to the controllers of DERs to be engaged, and if applicable, to determine the level of their production.

In order to increase the resilience of the MG, and to ensure a continuous and uninterruptible power supply, the MG must be able to meet the maximum demand of the load. Several aspects come into play, such as weather forecasts, the correct sizing of equipment and the appropriate optimisation methods. The management of RERs depends primarily on weather forecasts, which reduces the efficiency of RERs and can lead to an imbalance in the demand/consumption balance. Several studies are, therefore, interested in developing new methods for predicting weather data more accurately [125]. In [126], Obara and El-Sayed have developed an algorithm for optimal operation of a composite microarray using numerical weather information. This prediction is, then, used to minimise the fuel consumption of the system by developing a genetic algorithm (GA). As a result, energy production was not 100% accurate, but the fuel consumption was reduced. In [127], Ricalde et al. introduced some methods of weather forecasting as a function of the time interval of standby periods. Authors used Artificial Neural Networks (ANN) as a good approximation for non-linear and stochastic models. The multi layer ANN architecture is developed and formed with the Levenberg-Marqurardt backpropagation algorithm. The advantage is that these methods can predict with a considered accuracy the meteorological data, even with a lack of data.

A system that forecasts the load demand and availability of energy resources of micro grids is designed in [128] by Jaganmohan et al. It consists of three levels, the first being short (daily), medium (seasonal) and long term (annual). The system adapts an ANN function to forecast at the same time the demand and availability of energy resources in different situations and at the scale of the previously mentioned three levels. The contribution is remarkable, especially at the annual level, as the method gained experience by following the evolution of the daily and monthly loads.

The EMS must be able to ensure the proper operating of the MG while minimising a cost function. The cost function differs from project to project. It includes the cost of operating the MG, the cost of degrading the ESSs, the cost of degrading the DGs, the cost of CO_2 emissions, the cost of losses, the cost of the energy transaction, and the maintenance cost. The latter is refereed to in the literature as system resilience and self-healing capability. Resilience refers to the ability of a system to react to unexpected events by isolating the elements that cause a problem while allowing the rest of the system to recover and return to its normal operating regime. These self-healing operations result in fewer service interruptions to consumers and help service providers to manage the electricity distribution infrastructure in the best possible way [129].

6.2. ESSs Constraints

The deployment of ESSs in MGs is essential to ensure continuity of supply to the load, and to increase the efficiency of the system. Three criteria are related to ESSs as discussed in [130]: their location, their sizing and their optimal operation. Location and sizing are

necessary to overcome peak loads. Moreover, with good operational management, ESSs facilitate the integration of DERs including RERs, and minimise MG expenses.

The required storage capacity of the ESS can be determined using the autonomy of each component in number of days (e.g., batteries), which means the maximum time a component can continuously provide energy without being recharged by the MG energy sources. Furthermore, the tolerated threshold of discharge level (depth of discharge) of each energy storage component is of paramount importance. System voltage is typically involved in ESS modeling for grid voltage regulation [131]. The calculation of the number of each ESS needed in the MG is based on the storage capacity of a single element, its efficiency, the efficiency of the converters connected to that element [131].

The cost of the ESSs is part of the initial investment of the MG, which confirms the importance of sizing all components. This cost is calculated as the sum of the single price and the annual maintenance cost [51]. The single price means the purchase price of the ESS in relation to its lifetime, as well as the cost of its installation. The purchase price of the ESSs and their installation is a variable value depending on the size of the ESS. The maintenance cost per year is also a variable cost proportional to the size of the ESS [51]. To size the ESS, the main objective function is to reduce the total cost function taking into consideration both the investment cost, te operating and maintenance costs, and the discount rate. Conversely, the objective function may be to maximise the total profit [51].

6.3. Software for EMS

Several software and programming languages are used for EMS implementation on MGs, including tools for simulating the operation of MG components, tools for sizing and siting, and other for solving optimization problems such as rule-based techniques or optimisation approaches.

An energy management simulation tool called WindSim is used in [132] and used for computational fluid dynamics for wind prediction and wind turbine power. It optimises the placement of wind turbines in onshore and offshore wind farms. In [133], authors have used a software package called PVsyst to simulate the efficiency of PVs and solar installation. PVsyst is designed to be used by engineers, researchers and architects for the design of MGs. PSCAD/EMTDC is used in [134] to simulate a suggested control strategy for MG. The advantage of PSCAD is that it simplifies construction, simulation and modelling, offering unlimited possibilities for electrical system simulation. It has a comprehensive library of system models, making it one of the preferred tools for researchers. The General Algebraic Modelling System (GAMS) is a modelling system with efficient optimizers to solve complex and large-scale mathematical programming and optimization problems [135]. In [136], a linear programming in GAMS is carried out by L. Majić et al. for economic studies of two MGs consisting of thermal and electrical loads and cogeneration units. Moreover, in [137] Chen et al. make an economic analysis, formulate an optimisation model and determine the optimal operating strategies for intelligent MG systems. The optimisation model is formulated as an integer programming model using GAMS and the resolution is done by the CPLEX optimiser. CPLEX is among the most efficient solvers of GAMS; CPLEX is useful for large-scale linear programming, integer programming and quadratic constraint problems. The outputs of GAMS and CPLEX provide optimal investment and operating solutions with cost reduction. Software called Versatile Energy Resource Allocation (VERA) is used in [138] by Khodaei to minimise the total cost of system planning. It also makes a prediction of demand coverage based on local weather conditions.

MATLAB is widely used for MG simulations and optimal operation design. The HOMER software is widely used for MG modelling purposes. In [139], Nayar et al. presented an innovative hybrid wind-electric/vehicle/diesel system implemented on three remote islands in the Republic of Maldives. Authors used HOMER to undertake the planning of the renewable energy system and to analyse the different options. It took into account the cost of the per unit of the consumed electricity, the fuel saved and the

initial capital requirements. This software simplifies the study of MGs, and has contributed greatly to the design of MGs, and is the most widely used software.

6.4. Optimization Methods for EMS Problems Solving

The term "optimisation" means a group of mathematical techniques oriented towards the selection of an optimal solution, while respecting the system criteria, among other available alternatives. Indeed, optimisation aims at calculating the best available values of a given objective function in a defined domain under a set of constraints, including a wide range of objective functions and domain types. Different disciplines are included in computational optimisation such as mathematics to formulate the model, operations research to model the system, computer science for algorithmic design and analysis, and software engineering to implement the models, and are introduced in [140].

The optimisation process is described as an iterative procedure, which essentially consists of an optimizer and a model [141]. For a given problem, the model identifies objectives, variables and constraints [142]. An iterative sequence continues between the optimizer and the model. The optimizer retrieves the model with a set of values of decision variables, while the model performs the calculation of the objective function and constraints. This information is used again by the optimizer to compute a new set of decision variables until a stopping criteria for the optimization algorithm are satisfied [141].

Computer optimization techniques include optimization algorithms, iterative and heuristic methods. The type of problem to be optimised decides which optimisation algorithm should be adopted. At the same time, there are many different classifications of optimisation problems, depending on the type of decision variables, objective functions and constraints. In [142], different categories are defined, such as: continuous and discrete, global and local, constrained and unconstrained, stochastic [143] and deterministic, multimodal and multi-objective, and heuristic and meta-heuristic optimisation approaches [144].

Sometimes an optimisation method may not find the optimal solution. Often, real systems are impracticable optimization problems due to the characteristics of the problem. For example, in a Linear Programming (LP) optimization problem, when all unknown variables must be integers, the problem becomes an Integer Linear Programming (ILP) or, Integer Programming (IP) problem. IP problems are, in many practical situations, Nondeterministic Polynomial-time hard (NP-hard) [145]. Unlike linear problems, which can be solved efficiently, NP-hard problems may require a very large computational time to obtain the optimum, resulting in exponential times for practical reasons. Thus, in recent years, the optimization community have proposed a lot of optimization approaches; such as approximate methods (including heuristic and metaheuristic approaches) to solve optimisation problems. Heuristic methods have been developed with the aim of choosing the right solution from a large set of solutions, that can be implemented with a minimum computational effort as optimisation techniques [146]. When classical optimisation techniques do not provide the optimal solution, heuristic approaches will be useful for optimisation problems. Metaheuristics are used to find an optimal solution from a discrete search space. The advantage of metaheuristics is that it can combine more than one heuristic method: the first is used to find a primary solution, and then a second can be used to find the best solution [106].

In general, the way of classifying metaheuristic algorithms is based on trajectory and population-based methods, but there are other classifications. In [106,147], authors have discussed most methods often used for optimization:

- Trajectory meta-heuristics, with main methods such as Simulated Annealing, Tabu Search, Greedy Randomized Adaptive Search Procedures, Variable Neighbourhood Search, and Iterated Local Search.
- Population-based meta-heuristics, suitable methods are described such as GA and particle swarm optimization (PSO),
- Bio-inspired metaheuristics, which are metaheuristics that imitate nature. Main methods are: Evolutionary algorithms, Swarm intelligence and Ecology-based algorithms.

In addition of that, other types of metaheuristics can be considered, such as hybrid metaheuristics, which combines other optimisation approaches, and parallel metaheuristics, which carries out several metaheuristic resolution in parallel [106]. For some very complex systems, no heuristic neither metaheuristic method can lead to precise solutions in short resolution times. Parallel computing is therefore an interesting way to obtain good solutions with short run times. Parallel calculation, as its name suggests, performs several calculations in parallel, such that large problems can be broken down into several sub-problems by performing many calculations, simultaneously. Common problems encountered in parallel computing for MG applications are Monte Carlo simulations [106] and dynamic programming [106].

The EMS is based on a mathematical method to compute the best solution for energy dispatching while satisfying grid and ESS constraints and production/consumption balance. Optimisation algorithms are therefore only mathematical techniques developed to solve complex problems. In the ideal case the problem is linear and easy to solve. However, in general the problems of optimising physical systems and especially MGs are non-linear systems that require a strong method to solve. Some the most suited methods for energy management of MGs are presented in the next subsections and advantages and disadvantages of these optimization methods are presented in Table 3.

6.4.1. Linear and Non-Linear Programming Methods

Non-linear optimisation solves optimisation problems whose model is non-linear, i.e., the equations defining the system are non-linear. A code based on linear programming for the energy management of a MG is adopted in [148] allowing optimal planning of the operation of DGs, and their optimal distribution while respecting the operational and economic constraints imposed by the purchase and sale of energy. Furthermore, it allows finding the optimal destination of controllable and non-controllable loads. In [149], an efficient algorithm is presented by Taha and Yasser, that is based on a predictive control model for an island MG. Their objectives are: cost reduction, electricity consumption and gas emissions reduction at the output of the DGs. They therefore have a multiobjective optimisation model with MILP. Vafaei and Kazerani used traditional optimisation techniques in [150] to select and size DERs and ESSs for a MG to minimise operational costs. The formulated optimisation model is a MIP (Mixed Integer Programming) problem in the GAMS environment. In [74], Daniel et al. determined the optimal functioning of the MG by using an extended evaluation and recourse horizon that allows a correct distribution of ESSs. To have a linear formulation, they decompose their problem into two stages, the first is a unit commitment problem and the second is an optimal power flow problem.

Gerro et al. [151] found a multi-objective hierarchical control solution based on an integrated cost and emission optimisation algorithm, adapted to the needs of small remote villages, from solar energy. A control loop is developed for the management of storage elements to notify consumers when prices are reduced by investing in increased storage capacity for the village microgrid. Hossein and Elnaz in [152] developed a stochastic model to increase profits and decrease imbalance costs by taking into account the uncertainties of wind turbine production, solar systems, consumption, market prices in the previous 24 h and imbalances. A new neural network method is used to predict the production of PVs and wind turbines. An energy management system in an interconnected MG is proposed by Tim et al. in [153]. The grid has been integrated with a photovoltaic system and the constraints must meet demand. The model is a centralised approach based on the concept of flexibility for end users. They use quadratic programming to achieve the most optimal economic allocation. Bahramirad et al. proposed in [154] a model of the optimal sizing of the energy storage system for both initial investment and expansion problems. The problem is addressed from an economic point of view, using a mixed integer programming (MIP) approach in order to optimise investment in the operating costs of ESSs and MGs. Mohsen et al. proposed in [155] a technique to distribute ESSs in grid-connected MGs to reduce the operating cost. Therefore, a multi-objective problem is formulated to generate the optimal charge/discharge activities of the storage based on consumption forecasts. Then, the costs of the individual MGs are added together to have a single cost to minimise. The problem becomes a single objective optimisation problem. From the non-linear model, an equivalent linear program without binary/integer variables is driven that is easy to solve.

Table 3. Advantages and disadvantages of optimization methods for EMS.

Methods	Advantages	Drawbacks	
Linear programming methods	It allows an optimal use of productive resources. It improves the quality of decisions by providing possible and practical solutions.	The objective function and the constraint equalities or inequalities must be linear, which is not always possible.	
Non-Linear programming methods	It relies on simplified techniques to solve complicated problems. It gives several possible optimal solutions, which is an advantage over the mixed-integer linear programming (MILP) formulation.	The computation is done in several iterations, and it is therefore computationally expensive.	
Heuristic methods	It performs decision making faster and simply through shortcuts and good calculations using rules of thumbs such as intelligent guessing, trial and error, process of elimination, past formulas and analysis of historical data.	The end result may not be the optimal solution, the decision made may be inaccurate and the data selected may be insufficient.	
Stochastic methods	It is completely explicit about the assumptions made, and it allows these assumptions to be tested using a number of techniques. As it models random variation in decision variables, it is possible to estimate the uncertainty of these variables and the optimal solution found.	It may be based on very simple and unrealistic assumptions. Its model is too computationally complex to implement and requires quite extensive statistical and computer skills than some simpler deterministic models.	
Dynamic programming methods	Dynamic programming divides the main problem into several less complex problems, which can be solved more easily from the smallest to the largest by keeping the intermediate solutions.	Solving problems recursively makes the process a bit complex.	
Fuzzy logic methods	The structures of fuzzy logic models are not complex, justifiable and robust because it does not require exact information sources. It can be programmed according to the circumstances in case of sensors failure.	The results are not always accurate. They are therefore perceived as depending on suspicions, sometimes this reasoning is confused with the probability hypothesis.	
Neural network methods	Artificial neural networks perform several calculations at once, can give results even with a lack of information thanks to its automatic learning process and its ability to generalize.	It needs processors with parallel processing power, according to their structure.	
Multi-agent systems methods	This approach increases the efficiency of the solution mainly through the application of negotiation rules, evaluation, and coordination.	This technique complicates a scheduling problem as it has to decompose criteria for each individual agent.	

6.4.2. Metaheuristic Methods

Heuristic algorithms are inspired by nature, they imitate the behaviour of living species (ants, bees, etc.) and collect current information to make a decision and find an optimal solution that must be verified afterwards, and it gives a prediction on how the next state of the event will be. Ogunjuyigbe et al. studied in [131] a multi-objective optimization, trying to reduce costs of the life cycle, dump energy and operating of the MG. The authors use a genetic algorithm to develop a technique to find the optimal location of renewable generation and ESSs in the MG. A MG consisting of wind, fuel cell for storage, DG and electrolyzer is analysed in [156]. The role of the fuel cell is to overcome the intermittency problem by providing energy when wind turbines are not cost-effective and to allow optimal operation of the DGs. They use a particle swarm optimisation algorithm that

allows the stack to operate when consumption is immense, to reduce the operating costs of the DG by 70%. In [157], Zhimin et al. proposed a new methodology based on GA to enable high integration of PVs in low voltage networks with batteries as storage element. Their objective functions are: minimising energy cost for customers and releasing distribution network constraints for distribution network operators. The proposed concept has been adopted by Western Power Distribution in the SoLa Bristol smart grid demonstration project. A very interesting idea is proposed in [158] by Kirthiga et al. by developing a methodology for transforming a distribution network into a MG capable of operating autonomously. The authors use PSO and GA to find the optimal sizing and sitting of distributed generators to make MGs autonomous. Other objectives are achieved such as the optimisation of losses and system costs, while respecting network constraints such as load and network constraints, and generator and balance constraints.

In [159], Gwo-Ching developed a quantum genetic algorithm to confirm the validity and accuracy of a mathematical model using real examples for a system containing PVs, wind turbines, water turbines, a fuel cell, gas turbine and a micro gas turbine taking into account different energy efficiencies of the DERs, the operation and maintenance costs and the emissions cost. A study on the optimisation of a MG is presented by Li et al. in [160]. They used the particle swarm algorithm, which can operate a MG in connected or isolated mode. Fluctuations in renewable sources and consumption demands in the MG are taken into account by the method, with appropriate predictions made one day in advance to eliminate these fluctuations. In [161], Fatemeh et al. used a backtracking search optimization algorithm to optimize their MG. The proposed algorithm quickly converges to the optimal solution and avoids local optima. Their objective is to track the uncertainties related to the grid DERs, which are PV, micro-turbine, wind, and battery, as well as the uncertainties related to the market and obviously the uncertainty in the load. In [162], authors developed an expert system for energy management for a MG containing PVs and wind turbines. The developed system predicts the energy production by wind turbine using neural networks. Then, based on these data, an improved fuzzy satisfying algorithm based on bacterial foraging is used in the EMS module to obtain the optimisation of the multi-objective problem.

6.4.3. Dynamic Programming Techniques

The operation of a stand-alone MG containing DGs, PVs and batteries is optimized in terms of operating cost and emission in [163] using a dynamic scheduling technique that takes into consideration system constraints such as load satisfaction and DER generation capacity. An algorithm based on dynamic programming is proposed in [164] for the management of clustered MGs. Authors used a deep learning algorithm running in real time to generate the daily schedule of the MG, ten instructions are sent to the local control units to make a centralized control. Julio et al. presented an EMS for an autonomous residence connected to the main grid based on the state of charge of the batteries [165]. A persistence forecasting method is adapted for solar irradiance and wind speed and load forecasting using the Meteogalicias THREDDS server for weather forecasting. The control of the residence according to the state of charge of the batteries helps to reduce the fluctuation of the energy flow between the residence and the main grid. In [166], Hang et al. presented a technique based on approximate dynamic programming for economic management of the MG. Authors take into consideration RER output over time, energy price and load as stochastic variables. The piecewise linear function approximation with an improved slope updating strategy is used for the proposed method.

In [167], authors developed an EMS based on a dynamic programming approach to reduce the cash flow including the price of energy exchange with the main grid and the cost of battery degradation for a MG connected to the main grid. In [168], authors aimed to increase the profit from the sale of renewable energy and reduce the cost of balancing demand and consumption. To this end, they used energy market data to generate battery control setpoints using a dynamic scheduling method. An EMS for a large building is

designed in [169] operating in real time. Authors developed a control algorithm for alternating between all modes of battery operation, charging, discharging and load shedding, and for limiting the output from PVs. They also controlled the DC bus to keep it constant using the uncertainty of the PVs and the energy demand.

6.4.4. Multi-Agent Systems

Multi-agent systems (MAS) is a collection of intelligent agents that interact to solve problems that cannot be solved by a single agent or system. For some time, architectures and designs have been proposed for applications in engineering models, in general. With the increasing use and modelling of DERs for MGs work, MAS are well placed to be adapted to handle the dimensions and complexity of these energy systems. In [170], authors have proposed a new mechanism that pushes customers to contribute to the grid management in order to reduce peak loads and give consumers a high cost-performance ratio. They used the JADE programming language to develop a multi-agent based grid management system taking into consideration the different load patterns and available energy of DERs. A recent simulation platform for EMSs is developed in [171] in the clientserver framework and made available in C^{++} . The authors designed a hybrid EMS for a multi-agent based MG that can work with both centralised and decentralised approaches to reduce the cost of MG operation. In [172], authors performed the sizing of a MG composed of PV, wind, DG and batteries using the multi-objective self-adaptive differential evolution algorithm. Due to the multi-task nature of the used technique, the computation time is reduced, such that each part deals with a specific task. Authors, then, developed the EMS of the MG system. The multi-objective optimisation approach developed is used to track the probability of power loss, the price of electricity and the renewable factor.

In [173], Bogaraj and Kanakaraj developed an adaptive scheme based on the concept of MAS for energy management in an isolated MG. The generation system is seen as two parts; a primary part containing the PVs and wind turbines, when they fail to satisfy the load the second part, which is the batteries intervenes to ensure the continuity of the load supply. They have also developed predictive models to give information on wind, irradiance and consumption. In the extreme case, where both parts of the MG are not able to satisfy the load, a load shedding of part of the load is performed according to a fixed priority. A STATCOM-based compensation is designed to provide the necessary reactive power and to minimise voltage fluctuations. In [174], the authors used a multi-agent management methodology based on the JADE differential evolution algorithm to optimize the power generation costs caused by the intermittency of the DERs and the unknown variation in the load shape, especially the critical loads, knowing that the grid consists of two MGs containing DERs, PVs, and wind turbines and a load for each of the networks. In [175], Sajad et al. presented a decentralized multi-agent based algorithm to manage possible power imbalance situations in a PV-based autonomous residence connected to the main grid. The approaches are: grid agents, storage agents and user agents. In [176], Karavas et al. used cognitive maps and fuzzy logic techniques to develop a decentralised EMS based on the multi-agents present in a microgrid. Intelligent agents refer to DERs and ESSs and also for electrolysis and its fuel cells. This decentralised approach has shown several benefits compared to the centralised approach especially during outages.

6.4.5. Stochastic Methods and Robust Programming

The authors in [177] rely on the Lyapunov method to generate an optimization algorithm to deal with the uncertainty due to the intermittency of RERs in an interconnected MG. The method consists of two parts, in the first part a DG is used, in the other part an economic dispatch of conventional and distributed generation is performed based on hourly marketing. In [178], authors presented an optimization of the operation of a grid-connected power generation system. They developed a stochastic scheduling model taking into account the uncertainties of RER and consumption. First, the scheduling model optimizes the primary investment of the MG and then it provides the operation schedule to ensure a reduced cost. In [179], Juan et al. discussed a new charging strategy for optimized operation in a system containing wind turbines, batteries and DG.

A stochastic EMS for frequency gap reduction is developed in [180] and used for economic management of a MG in terms of operation and RERs reserves. The developed approach has been tested for challenging situations and has shown its effectiveness. In [181], authors presented a hierarchical management methodology to monitor and control the energy flow in the MG, i.e., DER generation and load consumption. A small-scale stochastic algorithm is proposed to overcome the uncertainty in the prediction of DERs. In [182], Wencong et al. presented some studies on a modified IEEE 37-bus test line. They formulated a model to reduce the operational cost of the microgrid and the power losses using one of the stochastic approaches. In [183], Taher et al. proposed a stochastic model based on teaching and learning to obtain the best Pareto optimal front. The objectives are cost and emissions minimisation. In this work, authors take into consideration the uncertainties related to the consumption demand, the available power output of DERs, as well as the energy cost in the market.

A stochastic coordination framework is introduced in the architecture of an agricultural MG to reduce the operating cost of the ESSs and lower the exchange cost with the main grid [184]. For weather forecasting, authors used a point estimation method. An EMS based on the productivity of the DERs and the unpredictable variation of the customers consumption is developed in [185]. The EMS is performed in two steps; first it schedules the network respecting the limits of the DERs in order avoid damaging its components. Then, it regulates the frequency by evaluating in real time the limit of energy capacity deviation. In [186], authors presented a stochastic multi-objective technique for managing a hybrid MG. The interest of this work is to reduce the network losses as well as the economic optimization of the RERs operation in the MG. The cost and losses model is developed based on weighting sum, and the solution was achieved by mixed integer linear programming.

6.4.6. Neural Network Techniques

In [187], the authors aim to make their network autonomous, they tried to maximise the exploitation of the RERs and reduce the carbon emissions by reducing the operation of conventional generators. Therefore, they developed an EMS model for their network, which operates in a connected mode. Neural networks are adopted in this work, one part for the management strategy and a second part to monitor the performance of the system in terms of battery degradation and the integration rate of the RERs. Cruz et al. designed an artificial neural network algorithm to run the EMS of a MG composed of PVs, wind turbines and a public load, the energy is stored in batteries [188]. The objective of the algorithm is to give the right instructions to the DERs in order to maintain maximum power generation. In [51], Chen et al. proposed to use time-series and neural feedback network techniques for cost-benefit analysis for the optimal sizing of an energy storage system in a MG. The aim is to predict solar radiation and wind speed. The main problem has the form of a MILP, which is solved in AMPL (A Modelling Language for Mathematical Programming). Authors also introduced a specific artificial neural network algorithm for the prediction of the production of DERs. Finally, they used a simple method for optimization.

In [189], Nnamdi et al. developed a game-theoretic demand response program to manage a MG. They were able to minimise the operating cost of the MG by increasing the output of DERs when fuel prices are low. In [190], Abir and Ali used neural networks to estimate the optimal tilt angle of PV at a given location to optimise the amount of energy produced by the PVs and thus reduce operating costs. A neural network method is used in [191], which consists of determining the optimal arrangement of power lines between micro-sources and load points. A traditional approach is, therefore, presented to design low-cost MGs architectures taking into account network reliability.

6.4.7. Fuzzy Logic Methods

In [192], authors introduced fuzzy logic techniques for the scheduling of storage devices. An adaptive billing price is set for the storage devices invoicing based on the price of local production of the MGs and the amount of the storage devices daily participation in the MGs. It applies a multi-purpose PSO method to find the right energy allocation for the MGs. In [193], Fossati et al. presented a fuzzy-based economic dispatch and MG unit commitment to optimise its energy management. It consists of two GAs; the first algorithm generates the network energy schedule and fuzzy rules and the second algorithm adjusts the fuzzy membership functions. In [194], Lydie et al. developed daily control rules to ensure a reliable grid containing a solar panel and a wind turbine as DERs, the load is assimilated to a residential demand. Taking into account the economic aspect, the control is based on fuzzy logic. In [195], the authors designed an EMS based on a controller containing more than 20 fuzzy logic properties to optimise the power flow in their MG by keeping the ESSs state of charge. In [196], authors have presented an optimal approach to voltage and frequency control using fuzzy logic. This approach has demonstrated high performance and desirable response for different load change scenarios.

In [197], a large-scale MIP model was developed to optimise the operating costs of an energy system (power system, district heating and its combinations). The main objective is to achieve reasonable runtimes and general applicability to all system situations. The method saves between 1 and 2% on operating costs. In [198], the authors designed an EMS to manage the energy flow tasks in a microgrid composed of RERs and ESSs. To achieve this optimization objective, they mixed a genetic algorithm with a Mamdani fuzzy logic algorithm. In [199], the authors presented a highly efficient EMS for the management of a connected network containing controllable and uncontrollable batteries and loads, with the objectives of reducing the power demand to the main grid while increasing the local generation of the RERs, based on a hybrid approach that combines wavelet functions and extended Kalman filtering to predict the consumption and production of RERs. In [200], Teo et al. proposed a methodology for managing ESSs tested against a rule-based control strategy. The simulation is done on MATLAB/Simulink and shows that this study increases the resilience of the MG.

6.4.8. Others Methods

The gradient method algorithm is used in [201] by selecting a power mix of four different types of MGs for economic allocation taking into account the penetration of renewable energy sources, associated costs and revenues. In [202], the authors used the Karush–Kuhn–Tucker (KKT) conditions to select the DERs to form a MG. The KKT method ensures that the true optimum is calculated. The KKT approach applied to non-linear programming generalises the Lagrange multiplier method, which only allows for equality constraints.

An algorithm for planning MGs is presented in [138] for the optimization of power generation and transmission. The objective is to minimize the total cost of MG planning. To do so, the problem is split into two sub-problems; a planning problem and an annual reliability problem. As the model is a non-linear model, the authors introduced a sequential quadratic programming (SQP) technique to fin an optimal solution. A probabilistic approach is proposed by Niknam et al. in [203]. Authors introduced a probabilistic uncertainty optimisation method and a modified multi-objective algorithm based on Modified Gravitational Search Algorithm to find the Pareto-optimal front for the economic/emissions management of MGs. In [204], Wishart et al. proposed a production system planning, with the objective of minimizing the total cost over a long term. The objective includes the reduction of line losses, reliability costs and initial investments. The Modified Discrete Particle Swarm Optimisation technique is used to optimise the problem by taking into account the system constraints: the supply current, the bus voltage and the DG output power.

In [205], authors have determined the optimal exploitation strategy and the optimisation scheme for the economic and environmental problem of a MG. They applied multi-objective optimisation based on modified game theory. The formulated model is constrained and non-linear, it takes into account NOx, SO_2 and CO_2 emissions. Comparisons are made with Multi-Objective Sequential Quadratic Sequential Programming, Multi-Objective Genetic Algorithms and Multi-Objective Direct Adaptive Mesh Search. The results demonstrate the effectiveness of the proposed approach in minimising the cost of operating the system while meeting the customers demand and system safety.

Table 4 describes the main contributions and some limitations of some energy management strategies presented in the literature.

Ref.	Methods	Contributions	Limitations		
[206]	Mixed integer non linear programming	Minimisation of general operating costs while maintaining the safety of the MG and ensuring their autonomy.	Two very important points are not taken into consideration: battery ESSs and the reduction of emission costs.		
[74]	Mixed integer non linear programming	The developed system can take into account the effects of grid imbalance and correct potential reactive power deficits.	Voltage limits are not taken into account. Detailed three-phase models are needed to overcome this problem.		
[207]	Non-Linear Programming	Demand mitigates the variability of renewable resources by allowing user demand to be controllable.	A robust optimisation method must be carried out in response to the demand to overcome the uncertainties of the microgrid.		
[153]	Non-Linear Programming	The developed method allows to optimally manage the use of the battery while minimizing the grid power.	The uncertainty of the PVs and the load are not taken into consideration.		
[156]	PSO algorithm	The operating costs of the DGs are well studied by selecting the right sizing and sitting.	Emissions minimisation is not considered. It must be taken into account in systems containing DGs.		
[208]	Evolutionary strategy	An efficient optimisation is presented that reduces the operating cost of PVs with batteries with an hourly variation in consumption considered in the study.	Seasonal and other types of consumption variations should be taken into account to obtain more accurate results.		
[209]	Particles Swarm Optimization	The wind uncertainty is taken into account as well as several important components such as PVs, wind turbines, battery bank, electrolyser, fuel cell and hydrogen tank.	The uncertainty of the wind leads to an increase in the cost, which must also be optimised.		
[210]	GA and MILP	The used technique is a very flexible set of sub-functions, an intelligent convergence behaviour, as well as diversified search approaches and penalty methods for constraint violations.	More parameters can be added in this approach to obtain a self-adopting system.		
[159]	Chaotic quantum genetic algorithm	The economic resolution by this method is efficient and presents interesting solutions.	Storage systems are not taken into account, neither is its uncertainty.		
[166,182]	Stochastic	The algorithm has a faster convergence rate. It effectively reduces the operational cost by taking into account the inherent intermittency and variability of renewable energy resources.	The proposed model can also be adapted to take into account other uncertainties such as load and customer behaviour.		

Table 4. Major papers dealing with energy management for MGs with focus on optimization approaches.

Ref.	Methods	Contributions	Limitations		
[211]	Multi-agent systems	This approach makes multi-agent systems well suited to the use and control of MGs. A step-by-step conceptual framework and platforms for the construction of multi-agent systems are developed.	Hardware incompatibility, the uncertainty inherent in the complexity of the software and the security risk for malicious external actors limit the use of management information systems for monitoring MGs.		
[196]	Fuzzy logic approach	The adapted fuzzy approach improved the coefficients of the PI voltage and frequency controllers.	This work is only one part of energy management of AC MGs. It must be adapted to deal with DC and hybrid microgrids.		
[200]	Fuzzy logic approach	The proposed methodology allows a control of the charge and discharge, which gives good results. The power consumption is effectively reduced.	A forecasting system is required to complete the model.		
[194]	Fuzzy logic approach	The results show that the load profile is well regulated by fuzzy logic rules.	Emissions should be taken into account to make the model more realistic. The ESSs also need to be inserted (batteries, for instance) allowing for islanded operation.		
[212]	Multi-agent system	Decision-making is facilitated by this method.	The approach is greatly complicated by the requirements for resilient, robust and rapid solutions.		
[190]	Artificial neural network	The technique is very effective, as it allows the optimal angle of inclination of pVs to be estimated with an accuracy of only 3 degrees.	This work is not an EMS but it can be integrated into EMS for more decision making.		
[183]	Stochastic programming	The most important advantage of this algorithm is the fast transfer of information between agents, allowing global optima to be found, even for complicated systems.	The only missing part is the storage system.		
[179]	Robust programming	The approach is useful for optimising the operation of wind-battery-diesel hybrid networks.	Optimisation of controllable load transfers is not assured.		

Table 4. Cont.

7. Future Research and Challenges in MGs

The elements allowing to improve the communication between the components of the MG, the security of the network and the use of artificial intelligence are research areas that require further investigation. Communication systems can be improved thanks to the development of the telecommunication field, wireless tools are undergoing very rapid development allowing much higher speeds and connecting by radio waves a greater number of devices (sensors, converters, control unit, modem Internet, etc.). The development of the various network components must be accompanied by the development of protection systems ensuring the stability of the network, avoiding the accidental destruction of expensive equipment, and data errors. Aspects of artificial intelligence are one of the promising solutions to overcome the problems encountered by the old methods such as: the lack of data (artificial intelligence adapts and can therefore continue without having the missing data), generation of an impractical solution (artificial intelligence understands the physical system, so it is able to recognize if the solution is applicable in reality), and finally it can be added in parallel with another method for more accuracy and speed.

7.1. Power-to-Gas Technology

The key element to improve the MGs resilience and ensure electricity supply is energy storage. Grid level-energy storage is mainly implemented using batteries. Unfortunately, batteries suffer from high investment cost, high degradation and replacement cost, and low cycle life. One technology that can be of great interest for the development of MGs is power-to-gas (P2G). P2G is a technology that uses electric power to produce hydrogen using electrolysis. When using surplus power from renewable resources, the gas is termed green-hydrogen.

Produced hydrogyn may be used as chemical raw material, or converted back into electricity using gas turbines. P2G allows energy from electricity to be stored and transported in the form of compressed gas, often using existing infrastructure for long-term transport and storage of natural gas. P2G is often considered the most promising technology for seasonal renewable energy storage [213].

7.2. Scalable Communication

Communication in MGs takes place through sensors installed at all network components. As the number of sensors increases with the integration of new devices, communication becomes more complicated and slow. Therefore, one of the interesting challenges is to develop a scalable communication infrastructure able to handle a very large amount of data and be able to perform new services and integrate other variables related to the new components inserted in the network [214].

Further research is needed to find more relevant tools to ensure good information transmission quality in MGs, targeting the overall operations of MGs, including the transient response of DERs. It is mandatory to reduce the time of sending sensitive information, which can be detected faults, switches and protection relays operating state in order to improve reliability. Control systems, such as voltage and reactive power control, are sometimes sensitive due to poor communication quality, and this problem needs to be studied. The communication between the MG devices and the MG architecture is realized according to the *IEC*60870-5-104 standard, it is important to move towards another standard *IEC*61850. This will allow a fast, reliable and secure access and control of substations, as well as a seamless interoperability [215]. The development of big data technologies, such as cloud computing, deep data mining, and machine learning methods, are necessary for the evolution of MGs. Specifically, these approaches allow processing the huge amount of data, RERs, power lines, and consumers bahaviour analysis, and uncertainty optimal management [216].

7.3. Cyber-Security Issues

Cyber-attacks are known to be the most common danger encountered in MGs, due to the vulnerability of the monitoring systems. These attacks can collapse the entire network, making it inoperable or endanger the confidentiality of employees and customer data. Attacks can mislead public services by generating false solutions in terms of utilisation capacity and hiding the attacks in progress. Future research must ensure confidentiality, authentication and privacy of information for network security and power delivery warranty.

Technically speaking, cyber-attacks against the MG causing infrastructure failures include cyber security flaws, cascading failures, blackouts, etc. It is therefore mandatory to strengthen the weak points with more coding. Currently, machine learning approaches are the most effective methods to detect and solve the problem of false information injection, by inserting logic meters for example. It is mandatory to further develop these approaches, and make them general so that they are adaptable to various scenarios as discussed in [214].

7.4. Machine Learning

Electricity distribution for networks heavily relying on intermittent generation sources can be compromised. In addition of being an additional constraint for electricity suppliers, this represents a major technical challenge to address. Indeed, highly intermittent production sources require more dynamic local control that can be based on various strategies such as interconnections with larger grids, load shedding, and energy storage. The main goal of local aggregator is, therefore, to guarantee the consistency of consumption and production data for real-time monitoring. Artificial intelligence can, then, be a performance tool for aggregators that allow promoting the penetration of renewable energy resources in the local energy mix [217]. Given the high costs of supplying non-interconnected areas such as island and rural areas, the implementation of smart MGs with the help of machine learning has a faster return on investment, which is conducive to experimentation on this type of network.

In addition, machine learning can be of great interest for electric vehicle charging infrastructure market, which is experiencing strong growth. Indeed, electric vehicle is an opportunity to develop large-scale intelligent charging solutions, both for public infrastructures and for private homes. Platforms could then control the charging of the fleet in accordance with the state of the network and consumers energy demand. Then, machine learning paves the way for more smart and reliable management of V2G technology. This would increase the flexibility of the network to cope with this new consumption technology.

8. Conclusions

This paper reviewed the main steps towards the design of microgrids and indexed the various challenges of their deployments. Several aspects have been addressed, from the selection of the appropriate installation site and the main DER and ESS for optimal operation, the optimal sizing of its components, to the types of control and EMS for an optimal MG operation schedule. The EMS is generally designed for optimal scheduling and efficient power distribution among the DERs, ESSs, and EVs. Several objectives can be selected to design an EMS, such as operational optimization, energy scheduling and resilience, and consideration of several environmental aspects, battery degradation, active demand response integration, line losses and system reliability, and consumer privacy. Thus, based on the desired objectives, the system is modelled, and from this model and its complexity, an optimization method is chosen to perform the solution respecting all the constraints of the chosen grid. A comprehensive and critical review is made of the energy management systems and solving approaches. The most known and used resolution tools are also presented. Uncertainty, related to renewable energy and random load variations, reduces the reliability of microgrids, which must be improved by using heuristic methods, for example to predict future events. Artificial intelligence aspects can be introduced in the future works of MGs optimal operation, in order to improve the components sizing, energy management system and location of renewable resources, considering the uncertainty of the load, weather data and the energy market. Controllable loads are the best solution to smooth the load curve and relieve DERs. Their integration should therefore be maximized by replacing all critical loads with controllable loads where possible.

A particular emphasis has been placed on the integration of EVs into microgrids. Specifically, the possibility of using the EVs fleet as distributed energy storage devices, that has been widely studied in the literature and deeply reviewed in this paper. This is because EVs can be charged using renewable energy resources during off-peak periods to significantly decrease the environmental impact of this type of transportation. The energy available on EV batteries can be used during on peak periods to support the grid and avoid congestion problems and their consequences. EVs have already been introduced in MGs to increase reliability and ensure continuity of generation. This concept has become an attractive topic in the electrical engineering and energy conversion research community, and needs further improvement to bring the expected benefits.

Author Contributions: Conceptualization, O.O., E.E. and Y.A.; methodology, O.O. and E.E.; validation, E.E. and Y.A. and E.S.G.; formal analysis, O.O., E.E. and Y.A.; investigation, O.O. and E.E.; resources, O.O. and E.E.; writing—original draft preparation, O.O.; writing—review and editing, E.E., Y.A. and E.S.G.; visualization, O.O.; supervision, E.E., Y.A. and E.S.G.; project administration, E.E.; funding acquisition, E.E. and Y.A. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by Brest Métropole, France.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

Abbreviations

The following abbreviations are used in this manuscript:

MG	Microgrid
EMS	Energy Management System
RER	Renewable Energy Resources
ESS	Energy Storage System
DG	Diesel Generator
DER	Distributed Energy Resources
EV	Electric Vehicle
V2G	Vehicle to Grid
P2G	Power to Gas
DC	Direct Current
AC	Alternative Current
PV	Photovoltaic Panel
LIB	Lithium Ion Battery
RFB	Redox Flow Battery
SIB	Na-ion Batteries
EC	Electrochemical Capacitors
FW	Flywheel
IEC	International Electrotechnical Commission
GA	Genetic Algorithm
ANN	Artificial Neural Networks
LP	Linear Programming
ILP	Integer Linear Programming
IP	Integer Programming
NP-hard	Nondeterministic Polynomial-time hard
HP	Heat pumps
FW	Flywheel
NWI	Numerical weather information
MIP	Mixed Integer Programming
MILP	Mixed integer linear programming
PSO	Particle Swarm Optimisation
IMP	Integer minimisation problem
ES	Evolutionary strategy
ACO	Ant Colony Optimisation
PAR	Peak-to-average ratio
SMES	Superconducting Magnetic Energy Storage
MAS	Multi-agent systems
KKT	Karush–Kuhn–Tucker
SQP	Sequential quadratic programming

References

- 1. ExxonMobil. Outlook for Energy; Technical Report; ExxonMobil: Irving, TX, USA, 2017.
- 2. Bull, S.R. Renewable energy today and tomorrow. Proc. IEEE 2001, 89, 1216–1226. [CrossRef]
- 3. Ellabban, O.; Abu-Rub, H.; Blaabjerg, F. Renewable energy resources: Current status, future prospects and their enabling technology. *Renew. Sustain. Energy Rev.* **2014**, *39*, 748–764. [CrossRef]
- 4. World Energy Council. World Energy Resources Marine Energy | 2016; Technical Report; World Energy Council: London, UK, 2016.
- 5. Ayamolowo, O.J.; Manditereza, P.; Kusakana, K. Exploring the gaps in renewable energy integration to grid. *Energy Rep.* **2020**, *6*, 992–999. [CrossRef]

- Farrokhabadi, M.; Cañizares, C.A.; Simpson-Porco, J.W.; Nasr, E.; Fan, L.; Mendoza-Araya, P.A.; Tonkoski, R.; Tamrakar, U.; Hatziargyriou, N.; Lagos, D.; et al. Microgrid stability definitions, analysis, and examples. *IEEE Trans. Power Syst.* 2019, 35, 13–29. [CrossRef]
- 7. Pannala, S.; Patari, N.; Srivastava, A.K.; Padhy, N.P. Effective Control and Management Scheme for Isolated and Grid Connected DC Microgrid. *IEEE Trans. Ind. Appl.* **2020**, *56*, 6767–6780. [CrossRef]
- Stadler, M.; Cardoso, G.; Mashayekh, S.; Forget, T.; DeForest, N.; Agarwal, A.; Schönbein, A. Value streams in microgrids: A literature review. *Appl. Energy* 2016, 162, 980–989. [CrossRef]
- 9. Lidula, N.; Rajapakse, A. Microgrids research: A review of experimental microgrids and test systems. *Renew. Sustain. Energy Rev.* **2011**, *15*, 186–202. [CrossRef]
- 10. Alhasnawi, B.N.; Jasim, B.H.; Siano, P.; Guerrero, J.M. A Novel Real-Time Electricity Scheduling for Home Energy Management System Using the Internet of Energy. *Energies* 2021, 14, 3191. [CrossRef]
- 11. K/bidi, F.; Damour, C.; Grondin, D.; Hilairet, M.; Benne, M. Power Management of a Hybrid Micro-Grid with Photovoltaic Production and Hydrogen Storage. *Energies* **2021**, *14*, 1628. [CrossRef]
- 12. Rashid, M.M.U.; Alotaibi, M.A.; Chowdhury, A.H.; Rahman, M.; Alam, M.S.; Hossain, M.A.; Abido, M.A. Home Energy Management for Community Microgrids Using Optimal Power Sharing Algorithm. *Energies* **2021**, *14*, 1060. [CrossRef]
- Parise, G.; Martirano, L.; Kermani, M.; Kermani, M. Designing a power control strategy in a microgrid using PID/fuzzy controller based on battery energy storage. In Proceedings of the 2017 IEEE International Conference on Environment and Electrical Engineering and 2017 IEEE Industrial and Commercial Power Systems Europe (EEEIC/I & CPS Europe), Milan, Italy, 6–9 June 2017; pp. 1–5.
- 14. Yoo, Y.; Al-Shawesh, Y.; Tchagang, A. Coordinated Control Strategy and Validation of Vehicle-to-Grid for Frequency Control. *Energies* **2021**, *14*, 2530. [CrossRef]
- 15. Latifi, M.; Sabzehgar, R.; Fajri, P.; Rasouli, M. A Novel Control Strategy for the Frequency and Voltage Regulation of Distribution Grids Using Electric Vehicle Batteries. *Energies* **2021**, *14*, 1435. [CrossRef]
- 16. Ota, Y.; Taniguchi, H.; Nakajima, T.; Liyanage, K.M.; Baba, J.; Yokoyama, A. Autonomous distributed V2G (vehicle-to-grid) satisfying scheduled charging. *IEEE Trans. Smart Grid* **2011**, *3*, 559–564. [CrossRef]
- 17. Han, S.; Han, S.; Sezaki, K. Development of an optimal vehicle-to-grid aggregator for frequency regulation. *IEEE Trans. Smart Grid* **2010**, *1*, 65–72.
- Shimizu, K.; Masuta, T.; Ota, Y.; Yokoyama, A. Load frequency control in power system using vehicle-to-grid system considering the customer convenience of electric vehicles. In Proceedings of the 2010 International Conference on Power System Technology, Hangzhou, China, 24–28 October 2010; pp. 1–8.
- Liyanage, K.; Yokoyama, A.; Ota, Y.; Nakajima, T.; Taniguchi, H. Impacts of communication delay on the performance of a control scheme to minimize power fluctuations introduced by renewable generation under varying v2g vehicle pool size. In Proceedings of the 2010 First IEEE International Conference on Smart Grid Communications, Gaithersburg, MD, USA, 4–6 October 2010; pp. 85–90.
- 20. Our World in Data. Evolution of Energy Consumption. Available online: https://ourworldindata.org/grapher/global-primaryenergy?country=OWID_WRL (accessed on 1 June 2020).
- 21. Notre-planete.info. Emissions Mondiales de CO₂ Dans l'atmosphère: Statistiques. Available online: https://www.notre-planete. info/indicateurs/CO2-emissions.phpL (accessed on 1 June 2020).
- 22. Zia, M.F.; Elbouchikhi, E.; Benbouzid, M. Microgrids energy management systems: A critical review on methods, solutions, and prospects. *Appl. Energy* **2018**, 222, 1033–1055. [CrossRef]
- 23. García Vera, Y.E.; Dufo-López, R.; Bernal-Agustín, J.L. Energy management in microgrids with renewable energy sources: A literature review. *Appl. Sci.* 2019, *9*, 3854. [CrossRef]
- Djelailia, O.; Necaibia, S.; Kelaiaia, M.S.; Merad, F.; Labar, H.; Chouial, H. Optimal Fuel Consumption Planning and Energy Management Strategy for a Hybrid Energy System with Pumped Storage. In Proceedings of the 2019 1st International Conference on Sustainable Renewable Energy Systems and Applications (ICSRESA), Tebessa, Algeria, 4–5 December 2019; pp. 1–6.
- 25. Marano, V.; Rizzoni, G. Energy and economic evaluation of PHEVs and their interaction with renewable energy sources and the power grid. In Proceedings of the 2008 IEEE International Conference on Vehicular Electronics and Safety, Columbus, OH, USA, 22–24 September 2008; pp. 84–89.
- 26. Habib, S.; Kamran, M.; Rashid, U. Impact analysis of vehicle-to-grid technology and charging strategies of electric vehicles on distribution networks—A review. *J. Power Source* 2015, 277, 205–214. [CrossRef]
- 27. Kempton, W.; Tomić, J. Vehicle-to-grid power implementation: From stabilizing the grid to supporting large-scale renewable energy. *J. Power Source* 2005, 144, 280–294. [CrossRef]
- 28. Clement-Nyns, K.; Haesen, E.; Driesen, J. The impact of vehicle-to-grid on the distribution grid. *Electr. Power Syst. Res.* 2011, *81*, 185–192. [CrossRef]
- 29. Zhang, Q.; Tezuka, T.; Ishihara, K.N.; Mclellan, B.C. Integration of PV power into future low-carbon smart electricity systems with EV and HP in Kansai Area, Japan. *Renew. Energy* **2012**, *44*, 99–108. [CrossRef]
- Pathomchaiwat, P.; Neammanee, B. Control of power management in the renewable energy system by using flywheel energy storage. In Proceedings of the 2017 14th International Conference on Electrical Engineering/Electronics, Computer, Telecommunications and Information Technology (ECTI-CON), Phuket, Thailand, 27–30 June 2017; pp. 513–516.

- Bharath, K.; Kodoth, R.; Kanakasabapathy, P. Application of Supercapacitor on a Droop-Controlled DC Microgrid for Surge Power Requirement. In Proceedings of the 2018 International Conference on Control, Power, Communication and Computing Technologies (ICCPCCT), Kerala, India, 23–24 March 2018; pp. 273–277.
- 32. Wang, B.; Sechilariu, M.; Locment, F. Intelligent DC microgrid with smart grid communications: Control strategy consideration and design. *IEEE Trans. Smart Grid* 2012, *3*, 2148–2156. [CrossRef]
- 33. Roy, A.; Auger, F.; Bourguet, S.; Dupriez-Robin, F.; Tran, Q.T. Intégration D'énergies Marines Renouvelables Dans un Système Multi-Sources. In Proceedings of the 3ème Symposium de Génie Electrique (SGE 2018), Nancy, France, 3–5 July 2018.
- Soofi, A.F.; Gharehpetian, G. Coordinated Droop Control of Battery and Flywheel ESSs in Isolated Microgrid Considering Their SOC. In Proceedings of the 2018 Smart Grid Conference (SGC), Piscataway, NJ, USA, 28–29 November 2018; pp. 1–5.
- 35. Dorin, P.; Toma, P.; Radu, E.; Eniko, L. Supplying a renewable energy single phase microgrid from a biomass generator using a three phase induction machine. In Proceedings of the 2016 10th International Conference on Compatibility, Power Electronics and Power Engineering (CPE-POWERENG), Bydgoszcz, Poland, 29 June–1 July 2016; pp. 208–213.
- Barin, A.; Canha, L.N.; Magnago, K.; da Rosa Abaide, A. Fuzzy multi-sets and multi-rules: Analysis of hybrid systems concerning renewable sources with conventional and flow batteries. In Proceedings of the 2009 15th International Conference on Intelligent System Applications to Power Systems, Curitiba, Brazil, 8–12 November 2009; pp. 1–6.
- 37. Hubble, A.H.; Ustun, T.S. Composition, placement, and economics of rural microgrids for ensuring sustainable development. *Sustain. Energy, Grids Netw.* **2018**, *13*, 1–18. [CrossRef]
- Zhang, Y.; Gatsis, N.; Giannakis, G.B. Robust energy management for microgrids with high-penetration renewables. *IEEE Trans.* Sustain. Energy 2013, 4, 944–953. [CrossRef]
- Morais, H.; Kádár, P.; Faria, P.; Vale, Z.A.; Khodr, H. Optimal scheduling of a renewable micro-grid in an isolated load area using mixed-integer linear programming. *Renew. Energy* 2010, 35, 151–156. [CrossRef]
- Carrasco, J.M.; Franquelo, L.G.; Bialasiewicz, J.T.; Galván, E.; PortilloGuisado, R.C.; Prats, M.M.; León, J.I.; Moreno-Alfonso, N. Power-electronic systems for the grid integration of renewable energy sources: A survey. *IEEE Trans. Ind. Electron.* 2006, 53, 1002–1016. [CrossRef]
- 41. Zhou, Z. Modeling and Power Control of a Marine Current Turbine System with Energy Storage Devices. Ph.D. Thesis, Université de Bretagne occidentale-Brest, Brest, France, 2014.
- 42. Jureczko, M.; Pawlak, M.; Mężyk, A. Optimisation of wind turbine blades. J. Mater. Process. Technol. 2005, 167, 463–471. [CrossRef]
- Xu, Z.; Ma, J.; Zhang, Y.; Wang, F. A Tracing and Comparison of the Development of International Hydropower Resources. In Proceedings of the 2017 International Conference on Smart Grid and Electrical Automation (ICSGEA), Hunan, China, 27–28 May 2017; pp. 114–121.
- 44. Perera, A.; Attalage, R.; Perera, K.; Dassanayake, V. A hybrid tool to combine multi-objective optimization and multi-criterion decision making in designing standalone hybrid energy systems. *Appl. Energy* **2013**, *107*, 412–425. [CrossRef]
- 45. Jahangiri, P.; Aliprantis, D.C. Distributed Volt/VAr control by PV inverters. *IEEE Trans. Power Syst.* **2013**, *28*, 3429–3439. [CrossRef]
- 46. Parashar, S.; Swarnkar, A.; Niazi, K.; Gupta, N. Operational Management of Grid Connected Microgrid with Responsive Loads. In Proceedings of the 2019 8th International Conference on Power Systems (ICPS), Jaipur, India, 20–22 December 2019; pp. 1–5.
- Aho, J.; Buckspan, A.; Laks, J.; Fleming, P.; Jeong, Y.; Dunne, F.; Churchfield, M.; Pao, L.; Johnson, K. A tutorial of wind turbine control for supporting grid frequency through active power control. In Proceedings of the 2012 American Control Conference (ACC), Montreal, QC, Canada, 27–29 June 2012; pp. 3120–3131.
- 48. Cong, R.G. An optimization model for renewable energy generation and its application in China: A perspective of maximum utilization. *Renew. Sustain. Energy Rev.* 2013, *17*, 94–103. [CrossRef]
- 49. Zia, M.F.; Elbouchikhi, E.; Benbouzid, M.; Guerrero, J.M. Energy management system for an islanded microgrid with convex relaxation. *IEEE Trans. Ind. Appl.* **2019**, *55*, 7175–7185. [CrossRef]
- Sasikala, S.P.; Jeong, G.H.; Yun, T.; Kim, S.O. A perspective on R&D status of energy storage systems in South Korea. *Energy* Storage Mater. 2019, 23, 154–158.
- 51. Chen, S.; Gooi, H.B.; Wang, M. Sizing of energy storage for microgrids. IEEE Trans. Smart Grid 2011, 3, 142–151. [CrossRef]
- Paul, D.; Ela, E.; Kirby, B.; Milligan, M. *The Role of Energy Storage with Renewable Electricity Generation*; Technical Report, Technical Report NREL/TP-6A2-47187; National Renewable Energy Laboratory: Golden, CO, USA, 2010.
- 53. Miller, J.R.; Simon, P. Electrochemical capacitors for energy management. Sci. Mag. 2008, 321, 651–652.
- Boyes, J.D.; Clark, N.H. Technologies for energy storage. Flywheels and super conducting magnetic energy storage. In Proceedings of the 2000 Power Engineering Society Summer Meeting (Cat. No. 00CH37134), Seattle, WA, USA, 16–20 July 2000; Volume 3, pp. 1548–1550.
- Gautam, M.; Rao, K.; Saxena, B.K. Reduction in liquid hydrogen by weight due to storage in different sizes of containers for varying period of time. In Proceedings of the 2017 International Conference on Technological Advancements in Power and Energy (TAP Energy), Kollam, India, 21–23 December 2017; pp. 1–6.
- Wang, J.; Jie, F.; Wang, L.; Li, P.; Ma, Y.; Chen, Y.; Han, G.; Chen, Z. Strategy on Interruptible Load Selection for Precise Load Shedding System of Source-Grid-Load Friendly Interaction System. In Proceedings of the 2019 IEEE 3rd Conference on Energy Internet and Energy System Integration (EI2), Changsha, China, 8–10 November 2019; pp. 1765–1769.

- Xing, S. Microgrid emergency control based on the stratified controllable load shedding optimization. In Proceedings of the International Conference on Sustainable Power Generation and Supply (SUPERGEN 2012), Hangzhou, China, 8–9 September 2012.
- 58. Gungor, V.C.; Sahin, D.; Kocak, T.; Ergut, S.; Buccella, C.; Cecati, C.; Hancke, G.P. Smart grid technologies: Communication technologies and standards. *IEEE Trans. Ind. Inform.* **2011**, *7*, 529–539. [CrossRef]
- 59. Kabalci, Y. A survey on smart metering and smart grid communication. Renew. Sustain. Energy Rev. 2016, 57, 302–318. [CrossRef]
- 60. Liao, J.T.; Huang, H.W.; Yang, H.T.; Li, D. Decentralized V2G/G2V scheduling of EV charging stations by considering the conversion efficiency of bidirectional chargers. *Energies* **2021**, *14*, 962. [CrossRef]
- 61. Khoucha, F.; Benbouzid, M.; Amirat, Y.; Kheloui, A. Integrated energy management of a plug-in electric vehicle in residential distribution systems with renewables. In Proceedings of the 2015 IEEE 24th International Symposium on Industrial Electronics (ISIE), Rio de Janeiro, Brazil, 3–5 June 2015; pp. 717–722.
- 62. Renault Communication, Pagecran. WHAT'S THE DEAL WITH V2G, OR VEHICLE-TO-GRID? Available online: https://easyelectriclife.groupe.renault.com/en/outlook/energy/whats-the-deal-with-v2g-or-vehicle-to-grid/ (accessed on 1 June 2020).
- Wu, C.; Mohsenian-Rad, H.; Huang, J.; Jatskevich, J. PEV-based combined frequency and voltage regulation for smart grid. In Proceedings of the 2012 IEEE PES Innovative Smart Grid Technologies (ISGT), Washington, DC, USA, 16–20 January 2012; pp. 1–6.
- 64. Galus, M.D.; Zima, M.; Andersson, G. On integration of plug-in hybrid electric vehicles into existing power system structures. *Energy Policy* **2010**, *38*, 6736–6745. [CrossRef]
- 65. Mets, K.; Verschueren, T.; Haerick, W.; Develder, C.; De Turck, F. Optimizing smart energy control strategies for plug-in hybrid electric vehicle charging. In Proceedings of the 2010 IEEE/IFIP Network Operations and Management Symposium Workshops, Seoul, Korea, 19–23 April 2010; pp. 293–299.
- 66. Kempton, W.; Kubo, T. Electric-drive vehicles for peak power in Japan. Energy Policy 2000, 28, 9–18. [CrossRef]
- 67. Saber, A.Y.; Venayagamoorthy, G.K. Plug-in vehicles and renewable energy sources for cost and emission reductions. *IEEE Trans. Ind. Electron.* **2010**, *58*, 1229–1238. [CrossRef]
- 68. Sortomme, E.; Hindi, M.M.; MacPherson, S.J.; Venkata, S. Coordinated charging of plug-in hybrid electric vehicles to minimize distribution system losses. *IEEE Trans. Smart Grid* 2010, 2, 198–205. [CrossRef]
- Bojrup, M.; Karlsson, P.; Alakula, M.; Simonsson, B. A dual purpose battery charger for electric vehicles. In Proceedings of the PESC 98 Record. 29th Annual IEEE Power Electronics Specialists Conference (Cat. No. 98CH36196), Fukuoka, Japan, 22–22 May 1998; Volume 1, pp. 565–570.
- Hamza, A.; Tahir, H.B.; Siraj, K.; Nasir, M. Hybrid AC/DC Microgrid for Residential Applications. In Proceedings of the 2019 IEEE Third International Conference on DC Microgrids (ICDCM), Shimane, Japan, 20–23 May 2019; pp. 1–5.
- Asplund, G. Le Réseau de Grand Transport Réconcilie Alternatif et Continu. Technical Report. Efficacité énergétique et Réseaux électriques. 2007. Available online: https://library.e.abb.com/public/5ccca8486fc4de42c1257305003ce195/22-27%202M733 _FRA72dpi.pdf (accessed on 1 June 2020).
- Shimomachi, K.; Hara, R.; Kita, H. Comparison between DC and AC microgrid systems considering ratio of DC load. In Proceedings of the 2015 IEEE PES Asia-Pacific Power and Energy Engineering Conference (APPEEC), Brisbane, QLD, Australia, 15–18 November 2015; pp. 1–4.
- 73. Bharath, K.; Krishnan, M.M.; Kanakasabapathy, P. A review on dc microgrid control techniques applications and trends. *Int. J. Renew. Energy Res. IJRER* **2019**, *9*, 1328–1338.
- 74. Olivares, D.E.; Cañizares, C.A.; Kazerani, M. A centralized energy management system for isolated microgrids. *IEEE Trans.* Smart Grid 2014, 5, 1864–1875. [CrossRef]
- 75. Hakuto, Y.; Tsuji, T.; Qi, J. Autonomous decentralized stabilizing control of DC microgrid. In Proceedings of the 2017 IEEE Second International Conference on DC Microgrids (ICDCM), Nuremburg, Germany, 27–29 June 2017; pp. 292–296.
- 76. Ferreira, R.A.; Braga, H.A.; Ferreira, A.A.; Barbosa, P.G. Analysis of voltage droop control method for dc microgrids with Simulink: Modelling and simulation. In Proceedings of the 2012 10th IEEE/IAS International Conference on Industry Applications, Fortaleza, CE, Brazil, 5–7 November 2012; pp. 1–6.
- Kukandeh, Y.R.; Kazemi, M.H. Microgrid control in islanding and connected mode. In Proceedings of the Electrical Engineering (ICEE), Iranian Conference on, Isfahan, Iran, 8–10 May 2018; pp. 1173–1178.
- R&D Inside. LES MICROGRIDS: UNE OPPORTUNITÉ DE VALEUR POUR LE GROUPE EDF. Technical Report; LES DOSSIERS DE LA R&D D'EDF. 2017. Available online: https://www.edf.fr/sites/default/files/Lot%203/CHERCHEURS/Publications/20 17.10_-_mag_rd_inside_microgrids.pdf (accessed on 1 June 2020).
- 79. Katiraei, F.; Iravani, M.R.; Lehn, P.W. Micro-grid autonomous operation during and subsequent to islanding process. *IEEE Trans. Power Deliv.* **2005**, *20*, 248–257. [CrossRef]
- 80. Guerrero, J.M.; Blaabjerg, F.; Zhelev, T.; Hemmes, K.; Monmasson, E.; Jemei, S.; Comech, M.P.; Granadino, R.; Frau, J.I. Distributed generation: Toward a new energy paradigm. *IEEE Ind. Electron. Mag.* **2010**, *4*, 52–64. [CrossRef]
- Hatziargyriou, N.; Van Cutsem, T.; Milanović, J.; Pourbeik, P.; Vournas, C.; Vlachokyriakou, O.; Kotsampopoulos, P.; Ramos, R.; Boemer, J.; Aristidou, P.; et al. *Contribution to Bulk System Control and Stability by Distributed Energy Resources Connected at Distribution Network*; Technical Report; IEEE: New York, NY, USA, 2017.

- 82. Mwasilu, F.; Justo, J.J.; Kim, E.-K.; Do, T.D.; Jung, J.-W. Electric vehicles and smart grid interaction: A review on vehicle to grid and renewable energy sources integration. *Renew. Sustain. Energy Rev.* **2014**, *34*, 501–516. [CrossRef]
- 83. Das, H.; Rahman, M.; Li, S.; Tan, C. Electric vehicles standards, charging infrastructure, and impact on grid integration: A technological review. *Renew. Sustain. Energy Rev.* 2020, 120, 109–618. [CrossRef]
- 84. Waraich, R.A.; Galus, M.D.; Dobler, C.; Balmer, M.; Andersson, G.; Axhausen, K.W. Plug-in hybrid electric vehicles and smart grids: Investigations based on a microsimulation. *Transp. Res. Part Emerg. Technol.* **2013**, *28*, 74–86. [CrossRef]
- 85. Hu, Z.; Song, Y.; Xu, Z.; Luo, Z.; Zhan, K.; Jia, L. Impacts and utilization of electric vehicles integration into power systems. *Proc. CSEE* **2012**, *32*, 1–10.
- Ma, Y.; Zhang, B.; Zhou, X. An overview on impacts of electric vehicles integration into distribution network. In Proceedings of the 2015 IEEE International Conference on Mechatronics and Automation (ICMA), Beijing, China, 2–5 August 2015; pp. 2065–2070.
- Sauer, D.U.; Kleimaier, M.; Glaunsinger, W. Relevance of energy storage in future distribution networks with high penetration of renewable energy sources. In Proceedings of the CIRED 2009-20th International Conference and Exhibition on Electricity Distribution-Part 1, IET, Prague, Czech Republic, 8–11 June 2009; pp. 1–4.
- 88. Yang, J.; Wang, M.; Zhang, Y.; Wang, D.; Ye, F. Applying power battery of electric vehicles for regulating peak in grid. *East China Electr. Power* **2010**, *38*, 1685–1687.
- 89. Sun, Y.; Liu, W.; Su, M.; Li, X.; Wang, H.; Yang, J. A unified modeling and control of a multi-functional current source-typed converter for V2G application. *Electr. Power Syst. Res.* **2014**, *106*, 12–20. [CrossRef]
- Ma, Y.; Zhang, B.; Zhou, X.; Gao, Z.; Wu, Y.; Yin, J.; Xu, X. An overview on V2G strategies to impacts from EV integration into power system. In Proceedings of the 2016 Chinese Control and Decision Conference (CCDC), IEEE, Yinchuan, China, 28–30 May 2016; pp. 2895–2900.
- Saber, A.Y.; Venayagamoorthy, G.K. Unit commitment with vehicle-to-grid using particle swarm optimization. In Proceedings of the 2009 IEEE Bucharest PowerTech, Bucharest, Romania, 28 June–2 July 2009; pp. 1–8.
- 92. Oubrahim, Z.; Choqueuse, V.; Amirat, Y.; Benbouzid, M.E.H. Disturbances classification based on a model order selection method for power quality monitoring. *IEEE Trans. Ind. Electron.* **2017**, *64*, 9421–9432. [CrossRef]
- 93. Orjuela, J.A.F. Intégration des Véhicules Electriques Dans le Réseau électrique Résidentiel: Impact sur le déséquilibre et stratégies V2G Innovantes. Ph.D. Thesis, Université de Grenoble, Grenoble, France, 2014.
- 94. Hussain, S.S.; Ustun, T.S.; Nsonga, P.; Ali, I. IEEE 1609 WAVE and IEC 61850 standard communication based integrated EV charging management in smart grids. *IEEE Trans. Veh. Technol.* 2018, 67, 7690–7697. [CrossRef]
- 95. Goel, S.; Sharma, R.; Rathore, A.K. A Review on Barrier and Challenges of Electric Vehicle in India and Vehicle to Grid Optimisation. *Transp. Eng.* 2021, *4*, 100057. [CrossRef]
- Kempton, W.; Tomic, J.; Letendre, S.; Brooks, A.; Lipman, T. Vehicle-to-Grid Power: Battery, Hybrid, and Fuel Cell Vehicles as Resources for Distributed Electric Power in California. 2001. Available online: https://escholarship.org/uc/item/5cc9g0jp (accessed on 1 June 2020).
- 97. Dubarry, M.; Devie, A.; McKenzie, K. Durability and reliability of electric vehicle batteries under electric utility grid operations: Bidirectional charging impact analysis. *J. Power Source* **2017**, *358*, 39–49. [CrossRef]
- Awadallah, M.A.; Venkatesh, B. Energy storage in distribution system planning and operation: Current status and outstanding challenges. *Can. J. Electr. Comput. Eng.* 2019, 42, 10–19.
- 99. Uddin, K.; Dubarry, M.; Glick, M.B. The viability of vehicle-to-grid operations from a battery technology and policy perspective. *Energy Policy* **2018**, *113*, 342–347. [CrossRef]
- 100. Birnie, D.P., III. Solar-to-vehicle (S2V) systems for powering commuters of the future. J. Power Source 2009, 186, 539–542. [CrossRef]
- Tulpule, P.J.; Marano, V.; Yurkovich, S.; Rizzoni, G. Economic and environmental impacts of a PV powered workplace parking garage charging station. *Appl. Energy* 2013, 108, 323–332. [CrossRef]
- Guille, C.; Gross, G. A conceptual framework for the vehicle-to-grid (V2G) implementation. *Energy Policy* 2009, 37, 4379–4390.
 [CrossRef]
- 103. Sovacool, B.K.; Hirsh, R.F. Beyond batteries: An examination of the benefits and barriers to plug-in hybrid electric vehicles (PHEVs) and a vehicle-to-grid (V2G) transition. *Energy Policy* **2009**, *37*, 1095–1103. [CrossRef]
- 104. Ekman, C.K. On the synergy between large electric vehicle fleet and high wind penetration–An analysis of the Danish case. *Renew. Energy* **2011**, *36*, 546–553. [CrossRef]
- Hossain, M.A.; Pota, H.R.; Hossain, M.J.; Blaabjerg, F. Evolution of microgrids with converter-interfaced generations: Challenges and opportunities. *Int. J. Electr. Power Energy Syst.* 2019, 109, 160–186. [CrossRef]
- 106. Gamarra, C.; Guerrero, J.M. Computational optimization techniques applied to microgrids planning: A review. *Renew. Sustain. Energy Rev.* **2015**, *48*, 413–424. [CrossRef]
- 107. Bukar, A.L.; Tan, C.W.; Lau, K.Y. Optimal sizing of an autonomous photovoltaic/wind/battery/diesel generator microgrid using grasshopper optimization algorithm. *Sol. Energy* **2019**, *188*, 685–696. [CrossRef]
- 108. Al-Falahi, M.D.; Jayasinghe, S.; Enshaei, H. A review on recent size optimization methodologies for standalone solar and wind hybrid renewable energy system. *Energy Convers. Manag.* **2017**, *143*, 252–274. [CrossRef]

- 109. Khatib, T.; Ibrahim, I.A.; Mohamed, A. A review on sizing methodologies of photovoltaic array and storage battery in a standalone photovoltaic system. *Energy Convers. Manag.* **2016**, *120*, 430–448. [CrossRef]
- 110. Bansal, A.K.; Kumar, R.; Gupta, R. Economic analysis and power management of a small autonomous hybrid power system (SAHPS) using biogeography based optimization (BBO) algorithm. *IEEE Trans. Smart Grid* **2013**, *4*, 638–648. [CrossRef]
- Kaabeche, A.; Diaf, S.; Ibtiouen, R. Firefly-inspired algorithm for optimal sizing of renewable hybrid system considering reliability criteria. Sol. Energy 2017, 155, 727–738. [CrossRef]
- 112. Dufo-López, R.; Bernal-Agustín, J.L. Design and control strategies of PV-Diesel systems using genetic algorithms. *Sol. Energy* **2005**, *79*, 33–46. [CrossRef]
- Yahiaoui, A.; Fodhil, F.; Benmansour, K.; Tadjine, M.; Cheggaga, N. Grey wolf optimizer for optimal design of hybrid renewable energy system PV-Diesel Generator-Battery: Application to the case of Djanet city of Algeria. *Sol. Energy* 2017, 158, 941–951. [CrossRef]
- 114. Mohammed, O.; Amirat, Y.; Benbouzid, M.; Feld, G. Optimal design and energy management of a hybrid power generation system based on wind/tidal/pv sources: Case study for the Ouessant French island. In *Smart Energy Grid Design for Island Countries*; Springer: Berlin/Heidelberg, Germany, 2017; pp. 381–413.
- 115. Islam, F.R.; Al Mamun, K.; Amanullah, M.T.O. Smart Energy Grid Design for Island Countries: Challenges and Opportunities; Springer: Berlin/Heidelberg, Germany, 2017.
- 116. Sinha, S.; Chandel, S. Review of software tools for hybrid renewable energy systems. *Renew. Sustain. Energy Rev.* 2014, 32, 192–205. [CrossRef]
- 117. Tozzi Jr, P.; Jo, J.H. A comparative analysis of renewable energy simulation tools: Performance simulation model vs. system optimization. *Renew. Sustain. Energy Rev.* **2017**, *80*, 390–398. [CrossRef]
- 118. Aziz, N.I.A.; Sulaiman, S.I.; Shaari, S.; Musirin, I.; Sopian, K. Optimal sizing of stand-alone photovoltaic system by minimizing the loss of power supply probability. *Sol. Energy* **2017**, *150*, 220–228. [CrossRef]
- Anam, F.; Sahito, A.A.; Shah, A. Comparison of AC and DC Microgrid Considering Solar-Wind Hybrid Renewable Energy System. *ESTIRJ* 2018, 4, 33–38.
- 120. Justo, J.J.; Mwasilu, F.; Lee, J.; Jung, J.W. AC microgrids versus DC microgrids with distributed energy resources: A review. *Renew. Sustain. Energy Rev.* 2013, 24, 387–405. [CrossRef]
- 121. Manandhar, U.; Ukil, A.; Jonathan, T.K.K. Efficiency comparison of DC and AC microgrid. In Proceedings of the 2015 IEEE Innovative Smart Grid Technologies-Asia (ISGT ASIA), Bangkok, Thailand, 3–6 November 2015; pp. 1–6.
- 122. Fregosi, D.; Ravula, S.; Brhlik, D.; Saussele, J.; Frank, S.; Bonnema, E.; Scheib, J.; Wilson, E. A comparative study of DC and AC microgrids in commercial buildings across different climates and operating profiles. In Proceedings of the2015 IEEE First International Conference on DC Microgrids (ICDCM), Atlanta, GA, USA, 7–10 June 2015; pp. 1–6.
- Slyke, D.V. The Heart of an Energy Management System. 2016. Available online: http://site.ieee.org/sas-pesias/files/2016/03/ SCADA_20150316_Slides.pdf (accessed on 1 June 2020).
- 124. Chen, C.; Duan, S.; Cai, T.; Liu, B.; Hu, G. Smart energy management system for optimal microgrid economic operation. *IET Renew. Power Gener.* 2011, *5*, 258–267. [CrossRef]
- 125. Voyant, C. Prédiction de Séries Temporelles de Rayonnement Solaire Global et de Production D'énergie Photovoltaïque à Partir de Réseaux de Neurones Artificiels. Ph.D. Thesis, Université Pascal Paoli, Corte, France, 2011.
- 126. Obara, S.y.; El-Sayed, A.G. Compound microgrid installation operation planning of a PEFC and photovoltaics with prediction of electricity production using GA and numerical weather information. *Int. J. Hydrogen Energy* **2009**, *34*, 8213–8222. [CrossRef]
- 127. Ricalde, L.J.; Cruz, B.; Catzin, G.; Alanis, A.Y.; Sanchez, E.N. Forecasting for smart grid applications with higher order neural networks. In Proceedings of the World Automation Congress 2012, Puerto Vallarta, Mexico, 24–28 June 2012; pp. 1–6.
- 128. Reddy, Y.J.; Kumar, Y.P.; Kumar, V.S.; Raju, K.P. Distributed ANNs in a layered architecture for energy management and maintenance scheduling of renewable energy HPS microgrids. In Proceedings of the 2012 International Conference on Advances in Power Conversion and Energy Technologies (APCET), Mylavaram, Andhra Pradesh, India, 2–4 August 2012; pp. 1–6.
- 129. Castermans, R.; Saelens, G.; Brechet, T. Quelles Sont les Perspectives D'evolution des Microgrid en Belgique? Etude des Apports de L'economie Collaboratvie au Secteur de L'energie. Ph.D. Thesis, Louvain School of Management, Ottignies-Louvain-la-Neuve, Belgium, 2018.
- 130. Das, C.K.; Bass, O.; Kothapalli, G.; Mahmoud, T.S.; Habibi, D. Overview of energy storage systems in distribution networks: Placement, sizing, operation, and power quality. *Renew. Sustain. Energy Rev.* **2018**, *91*, 1205–1230. [CrossRef]
- Ogunjuyigbe, A.; Ayodele, T.; Akinola, O. Optimal allocation and sizing of PV/Wind/Split-diesel/Battery hybrid energy system for minimizing life cycle cost, carbon emission and dump energy of remote residential building. *Appl. Energy* 2016, 171, 153–171. [CrossRef]
- Proietti, S.; Sdringola, P.; Castellani, F.; Astolfi, D.; Vuillermoz, E. On the contribution of renewable energies for feeding a high altitude Smart Mini Grid. *Appl. Energy* 2017, 185, 1694–1701. [CrossRef]
- Couty, P.; Lalou, M.J.; Cuony, P.; Cotture, S.; Saade, V. Positive energy building with PV facade production and electrical storage designed by the Swiss team for the US Department of Energy Solar Decathlon 2017. *Energy Procedia* 2017, 122, 919–924. [CrossRef]
- Ahamed, M.; Dissanayake, U.; De Silva, H.; Pradeep, H.; Lidula, N. Modelling and simulation of a solar PV and battery based DC microgrid system. In Proceedings of the 2016 International Conference on Electrical, Electronics, and Optimization Techniques (ICEEOT), Chennai, India, 3–5 March 2016; pp. 1706–1711.

- 135. Rosenthal, R. GAMS: A User's Guide. Available online: https://www.gams.com/dd/docs/bigdocs/GAMSUsersGuide.pdf (accessed on 1 June 2020).
- 136. Majić, L.; Krželj, I.; Delimar, M. Optimal scheduling of a CHP system with energy storage. In Proceedings of the 2013 36th International Convention on Information and Communication Technology, Electronics and Microelectronics (MIPRO), Opatija, Croatia, 20–24 May 2013; pp. 1253–1257.
- 137. Chen, Y.H.; Lu, S.Y.; Chang, Y.R.; Lee, T.T.; Hu, M.C. Economic analysis and optimal energy management models for microgrid systems: A case study in Taiwan. *Appl. Energy* **2013**, *103*, 145–154. [CrossRef]
- 138. Khodaei, A.; Shahidehpour, M. Microgrid-based co-optimization of generation and transmission planning in power systems. *IEEE Trans. Power Syst.* 2012, *28*, 1582–1590. [CrossRef]
- Nayar, C.; Tang, M.; Suponthana, W. Wind/PV/diesel micro grid system implemented in remote islands in the Republic of Maldives. In Proceedings of the 2008 IEEE International Conference on Sustainable Energy Technologies, Singapore, 24–27 November 2008; pp. 1076–1080.
- 140. Banos, R.; Manzano-Agugliaro, F.; Montoya, F.; Gil, C.; Alcayde, A.; Gómez, J. Optimization methods applied to renewable and sustainable energy: A review. *Renew. Sustain. Energy Rev.* **2011**, *15*, 1753–1766. [CrossRef]
- 141. Diwekar, U.M. Introduction to Applied Optimization; Springer Nature: Berlin/Heidelberg, Germany, 2020; Volume 22.
- 142. Nocedal, J.; Wright, S. Numerical Optimization; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2006.
- 143. Fouskakis, D.; Draper, D. Stochastic optimization: A review. Int. Stat. Rev. 2002, 70, 315–349. [CrossRef]
- 144. Bianchi, L.; Dorigo, M.; Gambardella, L.M.; Gutjahr, W.J. A survey on metaheuristics for stochastic combinatorial optimization. *Nat. Comput.* **2009**, *8*, 239–287. [CrossRef]
- Kogan, K.; Nikolenko, S.; Keshav, S.; López-Ortiz, A. Efficient demand assignment in multi-connected microgrids with a shared central grid. In Proceedings of the 2013 Sustainable Internet and ICT for Sustainability (SustainIT), Palermo, Italy, 30–31 October 2013; pp. 1–5.
- 146. Erdinc, O.; Uzunoglu, M. Optimum design of hybrid renewable energy systems: Overview of different approaches. *Renew. Sustain. Energy Rev.* **2012**, *16*, 1412–1425. [CrossRef]
- 147. Binitha, S.; Sathya, S.S. A survey of bio inspired optimization algorithms. Int. J. Soft Comput. Eng. 2012, 2, 137–151.
- Delgado, C.; Domínguez-Navarro, J.A. Optimal design of a hybrid renewable energy system. In Proceedings of the 2014 Ninth International Conference on Ecological Vehicles and Renewable Energies (EVER), Monte-Carlo, Monaco, 25–27 March 2014; pp. 1–8.
- 149. Taha, M.S.; Yasser, A.R.M. Robust MPC-based energy management system of a hybrid energy source for remote communities. In Proceedings of the 2016 IEEE Electrical Power and Energy Conference (EPEC), Ottawa, ON, Canada, 12–14 October 2016; pp. 1–6.
- 150. Vafaei, M.; Kazerani, M. Optimal unit-sizing of a wind-hydrogen-diesel microgrid system for a remote community. In Proceedings of the 2011 IEEE Trondheim Powertech, Trondheim, Norway, 19–23 June 2011; pp. 1–7.
- 151. Prinsloo, G.; Mammoli, A.; Dobson, R. Discrete cogeneration optimization with storage capacity decision support for dynamic hybrid solar combined heat and power systems in isolated rural villages. *Energy* **2016**, *116*, 1051–1064. [CrossRef]
- 152. Shayeghi, H.; Shahryari, E. Integration and management technique of renewable energy resources in microgrid. In *Energy Harvesting and Energy Efficiency*; Springer: Berlin/Heidelberg, Germany, 2017; pp. 393–421.
- 153. Paul, T.G.; Hossain, S.J.; Ghosh, S.; Mandal, P.; Kamalasadan, S. A quadratic programming based optimal power and battery dispatch for grid-connected microgrid. *IEEE Trans. Ind. Appl.* **2017**, *54*, 1793–1805. [CrossRef]
- 154. Bahramirad, S.; Reder, W.; Khodaei, A. Reliability-constrained optimal sizing of energy storage system in a microgrid. *IEEE Trans. Smart Grid* 2012, *3*, 2056–2062. [CrossRef]
- 155. Sandgani, M.R.; Sirouspour, S. Coordinated optimal dispatch of energy storage in a network of grid-connected microgrids. *IEEE Trans. Sustain. Energy* **2017**, *8*, 1166–1176. [CrossRef]
- 156. Nivedha, R.R.; Singh, J.G.; Ongsakul, W. PSO based economic dispatch of a hybrid microgrid system. In Proceedings of the 2018 International Conference on Power, Signals, Control and Computation (EPSCICON), Thrissur, India, 6–10 January 2018; pp. 1–5.
- 157. Wang, Z.; Gu, C.; Li, F. Flexible operation of shared energy storage at households to facilitate PV penetration. *Renew. Energy* **2018**, *116*, 438–446. [CrossRef]
- 158. Kirthiga, M.V.; Daniel, S.A.; Gurunathan, S. A methodology for transforming an existing distribution network into a sustainable autonomous micro-grid. *IEEE Trans. Sustain. Energy* **2012**, *4*, 31–41. [CrossRef]
- 159. Liao, G.C. Solve environmental economic dispatch of Smart MicroGrid containing distributed generation system–Using chaotic quantum genetic algorithm. *Int. J. Electr. Power Energy Syst.* 2012, 43, 779–787. [CrossRef]
- 160. Li, H.; Eseye, A.T.; Zhang, J.; Zheng, D. Optimal energy management for industrial microgrids with high-penetration renewables. *Prot. Control. Mod. Power Syst.* 2017, 2, 1–14. [CrossRef]
- 161. Najibi, F.; Niknam, T.; Kavousi-Fard, A. Optimal stochastic management of renewable MG (micro-grids) considering electrothermal model of PV (photovoltaic). *Energy* 2016, 97, 444–459. [CrossRef]
- Motevasel, M.; Seifi, A.R. Expert energy management of a micro-grid considering wind energy uncertainty. *Energy Convers.* Manag. 2014, 83, 58–72. [CrossRef]
- 163. Luu, N.A.; Tran, Q.T.; Bacha, S. Optimal energy management for an island microgrid by using dynamic programming method. In Proceedings of the 2015 IEEE Eindhoven PowerTech, Eindhoven, The Netherlands, 29 June–2 July 2015; pp. 1–6.

- Wu, N.; Wang, H. Deep learning adaptive dynamic programming for real time energy management and control strategy of micro-grid. J. Clean. Prod. 2018, 204, 1169–1177. [CrossRef]
- 165. Pascual, J.; Barricarte, J.; Sanchis, P.; Marroyo, L. Energy management strategy for a renewable-based residential microgrid with generation and demand forecasting. *Appl. Energy* **2015**, *158*, 12–25. [CrossRef]
- 166. Shuai, H.; Fang, J.; Ai, X.; Tang, Y.; Wen, J.; He, H. Stochastic optimization of economic dispatch for microgrid based on approximate dynamic programming. *IEEE Trans. Smart Grid* **2018**, *10*, 2440–2452. [CrossRef]
- An, L.N.; Quoc-Tuan, T. Optimal energy management for grid connected microgrid by using dynamic programming method. In Proceedings of the 2015 IEEE Power & Energy Society General Meeting, Denver, CO, USA, 26–30 July 2015; pp. 1–5.
- 168. Zhuo, W. Microgrid energy management strategy with battery energy storage system and approximate dynamic programming. In Proceedings of the 2018 37th Chinese Control Conference (CCC), Wuhan, China, 25–27 July 2018; pp. 7581–7587.
- Sechilariu, M.; Wang, B.; Locment, F. Building-integrated microgrid: Advanced local energy management for forthcoming smart power grid communication. *Energy Build.* 2013, 59, 236–243. [CrossRef]
- 170. Nunna, H.K.; Doolla, S. Energy management in microgrids using demand response and distributed storage—A multiagent approach. *IEEE Trans. Power Deliv.* 2013, 28, 939–947. [CrossRef]
- Mao, M.; Jin, P.; Hatziargyriou, N.D.; Chang, L. Multiagent-based hybrid energy management system for microgrids. *IEEE Trans.* Sustain. Energy 2014, 5, 938–946. [CrossRef]
- 172. Ramli, M.A.; Bouchekara, H.; Alghamdi, A.S. Optimal sizing of PV/wind/diesel hybrid microgrid system using multi-objective self-adaptive differential evolution algorithm. *Renew. Energy* **2018**, *121*, 400–411. [CrossRef]
- 173. Bogaraj, T.; Kanakaraj, J. Intelligent energy management control for independent microgrid. Sādhanā 2016, 41, 755–769. [CrossRef]
- 174. Raju, L.; Morais, A.A.; Rathnakumar, R.; Ponnivalavan, S.; Thavam, L. Micro-grid grid outage management using multi-agent systems. In Proceedings of the 2017 Second International Conference on Recent Trends and Challenges in Computational Models (ICRTCCM), Tamilnadu, India, 3–4 February 2017; pp. 363–368.
- 175. Ghorbani, S.; Rahmani, R.; Unland, R. Multi-agent autonomous decision making in smart micro-grids' energy management: A decentralized approach. In *German Conference on Multiagent System Technologies*; Springer: Berlin/Heidelberg, Germany, 2017; pp. 223–237.
- 176. Karavas, C.S.; Kyriakarakos, G.; Arvanitis, K.G.; Papadakis, G. A multi-agent decentralized energy management system based on distributed intelligence for the design and control of autonomous polygeneration microgrids. *Energy Convers. Manag.* 2015, 103, 166–179. [CrossRef]
- 177. Hu, W.; Wang, P.; Gooi, H.B. Towards optimal energy management of microgrids with a realistic model. In Proceedings of the 2016 Power Systems Computation Conference (PSCC), Genoa, Italy, 20–24 June 2016; pp. 1–7.
- Hu, M.C.; Lu, S.Y.; Chen, Y.H. Stochastic programming and market equilibrium analysis of microgrids energy management systems. *Energy* 2016, 113, 662–670. [CrossRef]
- Lujano-Rojas, J.M.; Monteiro, C.; Dufo-López, R.; Bernal-Agustín, J.L. Optimum load management strategy for wind/diesel/battery hybrid power systems. *Renew. Energy* 2012, 44, 288–295. [CrossRef]
- 180. Rezaei, N.; Kalantar, M. Stochastic frequency-security constrained energy and reserve management of an inverter interfaced islanded microgrid considering demand response programs. *Int. J. Electr. Power Energy Syst.* **2015**, *69*, 273–286. [CrossRef]
- Zachar, M.; Daoutidis, P. Energy management and load shaping for commercial microgrids coupled with flexible building environment control. J. Energy Storage 2018, 16, 61–75. [CrossRef]
- 182. Su, W.; Wang, J.; Roh, J. Stochastic energy scheduling in microgrids with intermittent renewable energy resources. *IEEE Trans. Smart Grid* **2013**, *5*, 1876–1883. [CrossRef]
- Niknam, T.; Azizipanah-Abarghooee, R.; Narimani, M.R. An efficient scenario-based stochastic programming framework for multi-objective optimal micro-grid operation. *Appl. Energy* 2012, 99, 455–470. [CrossRef]
- Ghasemi, A. Coordination of pumped-storage unit and irrigation system with intermittent wind generation for intelligent energy management of an agricultural microgrid. *Energy* 2018, 142, 1–13. [CrossRef]
- 185. Liu, J.; Chen, H.; Zhang, W.; Yurkovich, B.; Rizzoni, G. Energy management problems under uncertainties for grid-connected microgrids: A chance constrained programming approach. *IEEE Trans. Smart Grid* **2016**, *8*, 2585–2596. [CrossRef]
- 186. Reddy, S.S. Optimization of renewable energy resources in hybrid energy systems. J. Green Eng. 2017, 7, 43-60. [CrossRef]
- 187. Venayagamoorthy, G.K.; Sharma, R.K.; Gautam, P.K.; Ahmadi, A. Dynamic energy management system for a smart microgrid. *IEEE Trans. Neural Netw. Learn. Syst.* 2016, 27, 1643–1656. [CrossRef]
- May, E.C.; Ricalde, L.; Atoche, E.; Bassam, A.; Sanchez, E. Forecast and Energy Management of a Microgrid with Renewable Energy Sources Using Artificial Intelligence. In *International Symposium on Intelligent Computing Systems*; Springer: Berlin/Heidelberg, Germany, 2018; pp. 81–96.
- Nwulu, N.I.; Xia, X. Optimal dispatch for a microgrid incorporating renewables and demand response. *Renew. Energy* 2017, 101, 16–28. [CrossRef]
- 190. Chatterjee, A.; Keyhani, A. Neural network estimation of microgrid maximum solar power. *IEEE Trans. Smart Grid* 2012, 3, 1860–1866. [CrossRef]
- Cui, Q.; Shu, J.; Zhang, X.; Zhou, Q. The application of improved BP neural network for power load forecasting in the island microgrid system. In Proceedings of the 2011 International Conference on Electrical and Control Engineering, Yichang, China, 16–18 September 2011; pp. 6138–6141.

- Mahmoud, T.S.; Habibi, D.; Bass, O. Fuzzy logic for smart utilisation of storage devices in a typical microgrid. In Proceedings of the 2012 International Conference on Renewable Energy Research and Applications (ICRERA), Nagasaki, Japan, 11–14 November 2012; pp. 1–6.
- Fossati, J.P.; Galarza, A.; Martín-Villate, A.; Echeverría, J.M.; Fontán, L. Optimal scheduling of a microgrid with a fuzzy logic controlled storage system. *Int. J. Electr. Power Energy Syst.* 2015, 68, 61–70. [CrossRef]
- 194. Roiné, L.; Therani, K.; Manjili, Y.S.; Jamshidi, M. Microgrid energy management system using fuzzy logic control. In Proceedings of the 2014 World Automation Congress (WAC), Waikoloa, HI, USA, 3–7 August 2014; pp. 462–467.
- 195. Prathyush, M.; Jasmin, E. Fuzzy logic based energy management system design for AC microgrid. In Proceedings of the 2018 Second International Conference on Inventive Communication and Computational Technologies (ICICCT), Coimbatore, India, 5 March 2018; pp. 411–414.
- Ahmadi, S.; Shokoohi, S.; Bevrani, H. A fuzzy logic-based droop control for simultaneous voltage and frequency regulation in an AC microgrid. Int. J. Electr. Power Energy Syst. 2015, 64, 148–155. [CrossRef]
- 197. Harhammer, P.; Schadler, A. Optimal energy management. In *System Modelling and Optimization*; Springer: Berlin/Heidelberg, Germany, 1986; pp. 304–311.
- De Santis, E.; Rizzi, A.; Sadeghian, A. Hierarchical genetic optimization of a fuzzy logic system for energy flows management in microgrids. *Appl. Soft Comput.* 2017, 60, 135–149. [CrossRef]
- 199. Urias, M.E.G.; Sanchez, E.N.; Ricalde, L.J. Electrical microgrid optimization via a new recurrent neural network. *IEEE Syst. J.* **2014**, *9*, 945–953. [CrossRef]
- Teo, T.T.; Logenthiran, T.; Woo, W.L.; Abidi, K. Fuzzy logic control of energy storage system in microgrid operation. In Proceedings of the 2016 IEEE Innovative Smart Grid Technologies-Asia (ISGT-Asia), Melbourne, Australia, 28 November–1 December 2016; pp. 65–70.
- 201. Augustine, N.; Suresh, S.; Moghe, P.; Sheikh, K. Economic dispatch for a microgrid considering renewable energy cost functions. In Proceedings of the 2012 IEEE PES Innovative Smart Grid Technologies (ISGT), Washington, DC, USA, 16–20 January 2012; pp. 1–7.
- Han, Y.; Young, P.; Zimmerle, D. Optimal selection of generators in a microgrid for fuel usage minimization. In Proceedings of the 2013 IEEE Power & Energy Society General Meeting, Vancouver, BC, Canada, 21–25 July 2013; pp. 1–5.
- 203. Niknam, T.; Golestaneh, F.; Malekpour, A.R. Probabilistic model of polymer exchange fuel cell power plants for hydrogen, thermal and electrical energy management. *J. Power Source* 2013, 229, 285–298. [CrossRef]
- Wishart, M.T.; Dewadasa, M.; Ziari, I.; Ledwich, G.; Ghosh, A. Intelligent distribution planning and control incorporating microgrids. In Proceedings of the 2011 IEEE Power and Energy Society General Meeting, Detroit, MI, USA, 24–28 July 2011; pp. 1–8.
- 205. Mohamed, F.A.; Koivo, H.N. Multiobjective optimization using modified game theory for online management of microgrid. *Eur. Trans. Electr. Power* **2011**, *21*, 839–854. [CrossRef]
- Helal, S.; Najee, R.; Hanna, M.; Shaaban, M.F.; Osman, A.; Hassan, M. An energy management system for hybrid microgrids in remote communities. In Proceedings of the 2017 IEEE 30th Canadian Conference on Electrical and Computer Engineering (CCECE), Windsor, ON, Canada, 30 April–3 May 2017; pp. 1–4.
- 207. Luo, Z.; Gu, W.; Wu, Z.; Wang, Z.; Tang, Y. A robust optimization method for energy management of CCHP microgrid. *J. Mod. Power Syst. Clean Energy* **2018**, *6*, 132–144. [CrossRef]
- 208. Logenthiran, T.; Srinivasan, D.; Khambadkone, A.; Raj, T.S. Optimal sizing of an islanded microgrid using evolutionary strategy. In Proceedings of the 2010 IEEE 11th International Conference on Probabilistic Methods Applied to Power Systems, Singapore, 14–17 June 2010; pp. 12–17.
- Navaeefard, A.; Tafreshi, S.M.; Barzegari, M.; Shahrood, A.J. Optimal sizing of distributed energy resources in microgrid considering wind energy uncertainty with respect to reliability. In Proceedings of the 2010 IEEE International Energy Conference, Manama, Bahrain, 18–22 December 2010; pp. 820–825.
- Nemati, M.; Braun, M.; Tenbohlen, S. Optimization of unit commitment and economic dispatch in microgrids based on genetic algorithm and mixed integer linear programming. *Appl. Energy* 2018, 210, 944–963. [CrossRef]
- 211. Kantamneni, A.; Brown, L.E.; Parker, G.; Weaver, W.W. Survey of multi-agent systems for microgrid control. *Eng. Appl. Artif. Intell.* **2015**, *45*, 192–203. [CrossRef]
- 212. Colson, C.; Nehrir, M.; Gunderson, R. Multi-agent microgrid power management. IFAC Proc. Vol. 2011, 44, 3678–3683. [CrossRef]
- 213. Götz, M.; Lefebvre, J.; Mörs, F.; Koch, A.M.; Graf, F.; Bajohr, S.; Reimert, R.; Kolb, T. Renewable Power-to-Gas: A technological and economic review. *Renew. Energy* **2016**, *85*, 1371–1390. [CrossRef]
- 214. Rivas, A.E.L.; Abrão, T. Faults in smart grid systems: Monitoring, detection and classification. *Electr. Power Syst. Res.* 2020, 189, 106602. [CrossRef]
- Bani-Ahmed, A.; Weber, L.; Nasiri, A.; Hosseini, H. Microgrid communications: State of the art and future trends. In Proceedings of the 2014 International Conference on Renewable Energy Research and Application (ICRERA), Milwaukee, WI, 19–22 October 2014; pp. 780–785.

- 216. Faheem, M.; Shah, S.B.H.; Butt, R.A.; Raza, B.; Anwar, M.; Ashraf, M.W.; Ngadi, M.A.; Gungor, V.C. Smart grid communication and information technologies in the perspective of Industry 4.0: Opportunities and challenges. *Comput. Sci. Rev.* 2018, 30, 1–30. [CrossRef]
- 217. Entreprise, T.; Picon, I. L'intelligence Artificielle Dans les Micro Grids énergétiques, Simple Utopie ou réalité de Demain? Available online: https://www.ecoledesponts.fr/sites/ecoledesponts.fr/files/documents/omnes_nicolas.pdf (accessed on 1 June 2020).